Does Trading Volume Mitigate or Amplify Mispricing?

Xinrui Duan Li Guo Jun Tu Luying Wang June 16, 2023

Abstract

We find that when volume is relatively low, trading volume is primarily driven by attention rather than disagreement. An increase in volume, reflecting heightened attention, can mitigate mispricing stemming from limited attention. In contrast, when volume is relatively high, we find a stronger correlation between volume and disagreement than attention. Here, an increase in volume, suggesting heightened disagreement, may enhance investor bias, thereby amplifying mispricing driven by such bias. Overall, whether trading volume amplifies or mitigates mispricing depends on both the volume state (high or low) and the source of mispricing, whether due to limited attention or investor bias.

Keywords: Trading volume, Mispricing, Limited Attention, Disagreement, Behavioral Bias

We thank Zhi Da, Kewei Hou, Dashan Huang, Roger, Loh, Lin Peng, Melvyn Teo, Hong Zhang, Guofu Zhou, and the participants of 2023 SMU Summer Camp for helpful comments and suggestions.

Duan: Shenzhen Audencia Financial Technology Institute, Shenzhen University, Shenzhen, Guangdong, China, 518060. Email: xrduan@szu.edu.cn.

Guo: School of Economics, Fudan University, Shanghai, China 200433. Email: guo li@fudan.edu.cn.

Tu: Corresponding author. Lee Kong Chian School of Business, Singapore Management University, 178899, Singapore; Email: tujun@smu.edu.sg.

Wang: Lee Kong Chian School of Business, Singapore Management University, 178899, Singapore; Email: luying.wang.2019@pbs.smu.edu.sg.

1. Introduction

Behavioral finance seeks to improve our understanding of financial data by employing models that incorporate assumptions about investors' beliefs, preferences, and cognitive limitations. Prominent findings link mispricing to two key forces: investor bias (e.g., extrapolative beliefs), and limited attention (Daniel, Hirshleifer, and Sun, 2020; Chen, He, Tao, and Yu, 2022).

Moreover, Han et al. (2022) have demonstrated that trading volume, when viewed as a measure of disagreement, can amplify mispricing. This is in line with Atmaz and Basak's (2018) suggestion that disagreement may intensify investor bias, thus leading to amplified mispricing, as observed in Han et al. (2022). On the other hand, Hou, Peng, and Xiong (2009) argue that trading volume, considered as an indicator of attention, can reduce certain forms of mispricing, such as earnings momentum, a notable mispricing anomaly attributable to limited attention.¹

This raises two crucial questions: First, does trading volume amplify or mitigate mispricing? Second, how to interpret trading volume as a measure of disagreement or attention, is it state-dependent? In this paper, we carry out an exhaustive investigation to address these questions and supply the associated economic explanations.

Empirically, we conduct a five-by-five independent double sort on post-earnings announcement drift (PEAD) and trading volume.² Our findings show that the monthly CAPM alpha of the PEAD long-short (L-S) portfolio decreases from 0.99% for low-volume (quintile 1) stocks to 0.10% for medium-volume (quintile 3) stocks, a difference of -0.89% (t-value = -3.81). It increases from 0.10% for medium-volume (quintile 3) stocks to 0.69% for high-volume (quintile 5) stocks, a difference of 0.59% (t-value = 2.77). Consequently, trading volume mitigates PEAD from quintiles 1 to 3 in the low-volume state and amplifies mispricing from quintiles 3 to 5 in the high-volume state.³

¹Chen, He, Tao, and Yu, (2022) document that the anomalies corresponding to many recently proposed anomaly-based prominent factors are less pronounced among firms with high investors' attention proxied by media coverage. Hence, as a potential proxy for investor attention, volume may weaken mispricing anomalies related to attention as documented in Chen, He, Tao, and Yu, (2022). Moreover, like these existing studies, we use "positive volume-return relation" and "volume amplification effect" interchangeably. We also use "negative volume-return relation" and "volume mitigation effect" interchangeably.

²Our results are robust using the portfolio sorting based on NYSE-breakpoints and sequential sorting methods. Please refer to the results in Appendix Table A3.

³ In this paper, we categorize stocks into low-volume state (quintiles 1-3) and high-volume state (quintiles 3-5).

Therefore, we have a U-shaped pattern of PEAD, which is mitigated by volume in the low-volume state while amplified by volume in the high-volume state.

To investigate the underlying mechanisms behind the U-shaped pattern of PEAD, we first show that while trading volume can capture both attention and disagreement, its ability to reflect these components is not fixed but state-dependent. In the low-volume state (quintiles 1-3 stocks), trading volume aligns more with attention than disagreement. Conversely, in the high-volume state (quintiles 3-5 stocks), it corresponds more with disagreement than attention. ⁴

Moreover, PEAD is generally more pronounced among firms with less investors' attention (Hou, Peng, and Xiong, 2009; Chen, He, Tao, and Yu, 2022). As such, investor inattention is high in quintile 1 stocks, resulting in substantial PEAD. As volume increases in the low-volume state from quintile 1 to 3, investor attention (inattention) dramatically increases (decreases), thereby reducing PEAD. On the other hand, in the high-volume state (quintiles 3-5 stocks), investor attention is high and mispricing in PEAD due to inattention is minimal. Yet, mispricing in PEAD due to expectation bias might still persist. Hence, an increase in volume implies increased disagreement, exacerbating expectation bias and amplifying PEAD.

In summary, our findings suggest that trading volume serves as a good indicator of disagreement when volume is high and of attention when volume is low. However, it does not act as a good indicator of disagreement when volume is low, nor of attention when volume is high. Therefore, whether to assign trading volume as a measure of either disagreement or attention is state-dependent, and it would be inappropriate to simply make this assignment. Furthermore, the effect of trading volume on mispricing depends on both the volume state and the origin of the mispricing, whether it's driven

Notably, stocks in quintile 3 belong to both the low-volume state and high-volume state categories.

⁴ We construct a composite attention measure based on the average ranking of six attention proxies, which include abnormal Google search volume, media coverage, abnormal EDGAR downloads, analyst coverage, price delay, and abnormal Bloomberg downloads. Moreover, we construct a composite disagreement measure based on the average ranking of three disagreement proxies, which include stock return volatility, analysts' return forecast dispersion, and analysts' earnings forecast dispersion.

⁵ Barberis, Shleifer, and Vishny (1998) propose an over-extrapolative ("trending") regime, which can be due to investors' biased extrapolative expectations about future earnings.

⁶ While excessive attention can intensify mispricing (Barber and Odean, 2008), the high-volume state maintains a consistent level of high attention. Thus, the amplification of PEAD from quintiles 3 to 5 is unlikely to be driven by this stable level of attention, regardless of its height.

by investor bias or limited attention. Specifically, in states of low trading volume, where volume is indicative of attention, it can mitigate mispricing caused by inattention. Conversely, in high-volume states, where volume is reflective of disagreement, it can amplify mispricing arising from bias.

Some anomalies, such as PEAD, primarily reveal investor inattention-driven mispricing in the low-volume state where attention is low. However, other anomalies may be less influenced by inattention and reflect mispricing due more to expectation bias rather than inattention-driven mispricing. An example is the FIN (financing) factor proposed by Daniel, Hirshleifer, and Sun (2020). After examining PEAD, we proceed to analyse the FIN anomaly.

We find that trading volume consistently amplifies FIN mispricing across all levels of trading volume. The monthly CAPM alpha of the FIN portfolio increases from 0.26% (quintile 1) to 0.63% (quintile 3), a difference of 0.37% (t-value = 1.27). It further increases to 1.04% (quintile 5) from 0.63% (quintile 3), with a difference of 0.40% (t-value = 1.53). The total increase from quintiles 1 to 5 is 0.77% (t-value = 2.02). Disagreement increases from quintiles 3 to 5, enhancing expectation bias and intensifying FIN. Since FIN reflects expectation bias rather than inattention-driven mispricing, the mitigation effect from quintiles 1 to 3 is largely muted. However, a significant rise in attention from quintiles 1 to 3 may magnify the effect of disagreement, inducing a volume amplification effect for FIN. Although a similar scenario might occur with the PEAD anomaly among quintiles 1 to 3, the mitigation effect tends to dominate the amplification effect in this low-volume state.

We further analyze the MISP (mispricing) factor of Stambaugh and Yuan (2017). As explained later in the paper, MISP, based on 11 anomalies, is likely significantly influenced by both inattention and expectation bias.⁷ In the low-volume state (quintiles 1 to 3), neither volume mitigation nor amplification effect is observed. This could be because the mitigation effect for inattention-driven mispricing and the amplification

⁷Among anomalies used in construction of MISP, some are affected by inattention, such as financial distress, Oscore, momentum, gross profitability, and return-on-assets (PERF) (Chen, He, Tao, and Yu, 2022), while other anomalies are affected by expectation bias, such as net share issuance (NSI) and composite share issuance (CSI) which also are used to construct FIN. Therefore, MISP includes the two types of mispricing captured by PEAD and FIN.

effect for expectation bias-driven mispricing in MISP are of similar magnitudes, thereby offsetting each other. However, in the high-volume state, a strong volume amplification is noted. This occurs because from quintiles 3 to 5, attention is consistently high with little change, making inattention-driven mispricing always relatively small with little change or reduction. Meanwhile, the expectation bias-driven mispricing could be amplified as disagreement significantly increases from quintiles 3 to 5.8

In addition, we apply Fama-MacBeth analysis to investigate how attention and disagreement might explain volume mitigation and amplification effects on mispricing. The findings are in line with the aforementioned double-sort approach. For example, when regressing one-month-ahead stock returns on PEAD, trading volume, and their interaction, we find a significant negative coefficient on the interaction between PEAD and trading volume, especially for quintiles 1 to 3. This implies that trading volume mitigates the positive prediction of PEAD on stock returns.

As trading volume might convey information beyond investor attention and disagreement, we explored three alternative explanations: arbitrage costs, illiquidity, and private information, using metrics such as IVOL, bid-ask spread, and volume-synchronized probability of informed trading (VPIN). Our findings are robust against these alternative explanations.

Finally, the state-dependent relationship between trading volume and either attention or disagreement is as follows: low attention can result in limited trading volume, as fewer investors are aware of certain stocks. As attention rises, more investors engage in trading, thereby increasing volume. However, if attention is already high, further increases may have a reduced impact on volume, making this relationship more pronounced in a low-volume state. As for disagreement, it can stimulate trading provided a certain level of attention exists for investors to notice it. When attention is low, a rise in disagreement among a smaller group of investors may only marginally increase trading volume. Conversely, when attention is high, the same level of

⁸Our results for MISP are consistent with the findings of Han et al. (2022) that trading volume amplifies mispricing captured by MISP which is mainly concentrated among stocks in high-volume state.

disagreement among a larger group can cause a significant volume increase, making this relationship stronger in a high-volume state.

The rest of the paper is organized as follows. Section 2 provides a brief review on the related literature and some discussions about the contributions of this study. Section 3 describes data and shows the summary statistics. Section 4 presents the main empirical results. Section 5 explores the underlying economic explanations. Section 6 concludes.

2. Literature Review and Contributions

Previous research has found that limited attention can explain many anomalies, such as post-earnings announcement drifts (Hirshleifer, Lim, and Teoh, 2009; DellaVigna and Pollet, 2009; Ben-Rephael, Da and Israelsen, 2017), momentum (Hong, Lim and Stein, 2000; Da, Gurun, and Warachka, 2014), and lead-lag return effects among economically-linked firms, such as customer-supplier links (Cohen and Frazzini, 2008), geographic links (Parsons, Sabbatucci, and Titman, 2020), competition links (Eisdorfer et al., 2022), product similarity (Hoberg and Phillips, 2010, 2016), shared analysts (Ali and Hirshleifer, 2020), among others.

Furthermore, investor bias has important implications for mispricing. One notable concept is the presence of extrapolative beliefs among individuals. This implies that investors' estimation of the future value of a quantity is a positive function of its recent past values. Several studies have utilized survey data to describe investors' extrapolative beliefs about future returns (Bacchetta et al., 2009; Amromin and Sharpe, 2014; Greenwood and Shleifer, 2014). For example, Greenwood and Shleifer (2014) used six survey databases to quantify investors' expectations of stock market returns and found a positive correlation between these expectations and past returns (i.e., extrapolation), while exhibiting a negative relationship with model-based expected returns (i.e., over-extrapolation). Daniel, Hirshleifer, and Sun (2020) motivate the financing factor (FIN) using extrapolative beliefs. Da et al. (2021) observed that individual investors tend to extrapolate from recent past returns, giving more weight to the most recent ones. Furthermore, Liao et al. (2022) applied the extrapolative model to explain the sharp increase in prices and volume observed during historical financial

bubbles.

Overall, a growing body of research proposes various behavioral approaches to explain mispricing or abnormal returns (Barberis, 2018). Inattention and investor bias are two critical factors that play important roles in explaining these phenomena.⁹

In terms of contributions, our study delves into the relationship between trading volume and mispricing, revealing that volume can either reduce mispricing due to increased attention in low-volume states or intensify mispricing driven by investor bias when volume signifies disagreement in high-volume states. This study offers explanations for the divergent outcomes observed in prior literature concerning trading volume's relation to mispricing. While some research indicates trading volume mitigates mispricing as a measure of investor attention (Hou, Peng, and Xiong, 2009), others suggest it amplifies mispricing reflecting belief dispersion (Han et al., 2022). Our findings underscore volume's dual role: both mitigating and amplifying different anomalies depending on whether the mispricing stems from limited attention or expectation bias. Unlike Hou, Peng, and Xiong (2009) who solely view volume as a proxy for attention, we consider it also indicative of disagreement.

Additionally, our study proposes a unique, state-dependent framework for attention and disagreement, the two components of trading volume. Given the challenge of directly measuring investor attention and disagreement, many studies use trading volume as a proxy. For instance, Gervais et al. (2001) suggest high trading volumes increase a stock's visibility, thus attracting more attention. Likewise, Hou, Peng, and Xiong (2009) use trading volume to measure investor attention. Lee and Swaminathan (2000) and Han et al. (2022) interpret trading volume as a measure of disagreement. However, we find that for low-volume state stocks, trading volume aligns closely with attention, whereas its connection with disagreement is less distinct. In contrast, for high-volume state stocks, trading volume correlates more strongly with disagreement than attention. These findings stress that while trading volume reflects both attention and disagreement, their respective weights are state-dependent. As such, caution is advised when using trading volume as a proxy for either attention or disagreement.

⁹There are some other important behavior factors, such as overconfidence.

Moreover, our study contributes to the literature on the trading volume-stock returns relationship. Despite extensive research, a clear understanding remains elusive (Chordia et al., 2001; Llorente et al., 2002; Lo et al., 2000; Israeli et al., 2022; Gervais et al., 2001; Kaniel et al., 2012). Notably, Chordia et al. (2001) find a negative relationship between stock returns and trading volume, while Gervais et al. (2001) observe a high-volume return premium. We propose that this discrepancy could be due to the intricate dynamics between trading volume and mispricing, contingent on both volume states and mispricing origins. For instance, in high-volume states, we find a positive volume-returns relationship for underpriced stocks, but a negative relationship for overpriced stocks when mispricing primarily results from investor bias. Conversely, in low-volume states, the volume-returns relationship reverses: it's negative for underpriced stocks and positive for overpriced stocks, when mispricing is typically driven by limited attention.

3. Data Description and Summary Statistics

This section outlines our methodology for constructing trading volume and our main anomaly variables, as well as proxies for investors' attention and disagreement. Our sample consists of all common stocks (CRSP share codes 10 and 11) listed on NYSE, AMEX, and NASDAQ (CRSP exchange codes 1, 2, and 3) from January 1980 to December 2021, with stocks priced below \$1 and stocks with negative book equity being excluded. We obtain stock return data from CRSP, firm accounting and financial statement variables from the merged CRSP-Compustat database, and analyst forecast data from I/B/E/S.

The key variable of interest is trading volume (TO), which is the average turnover over the prior twelve months. The monthly turnover is the number of shares traded during a month divided by the number of shares outstanding at the end of the month. We follow Gao and Ritter (2010) to adjust the volume of Nasdaq stocks. Specifically, we divide the volume by 2.0 before January 2001, by 1.8 for the rest of 2001, by 1.6 for 2002 to 2003, and leave it unchanged afterward.

The main anomaly variables include the short-horizon mispricing factor (PEAD) and long-horizon mispricing factor (FIN) from Daniel, Hirshleifer, and Sun (2020) and the

mispricing factor (MISP) from Stambaugh and Yuan (2017). The PEAD is measured as the 4-day cumulative abnormal return around the most recent quarterly earnings announcement date following Chan, Jegadeesh, and Lakonishok (1996) and Daniel, Hirshleifer and Sun (2020). The FIN is the financing factor based on the 1-year net share issuance (NSI) and 5-year composite share issuance (CSI) measures from Pontiff and Woodgate (2008) and Daniel and Titman (2006), respectively. We follow Daniel, Hirshleifer and Sun (2020)'s method to sort the firms into three financing groups (low "L", middle "M", or high "H") based on NSI and CSI rankings of each month. The MISP is the composite mispricing factor obtained by averaging rankings of 11 anomalies, including net stock issues, composite equity issues, accruals, net operating assets, asset growth, and investment to assets, financial distress, O-score, momentum, gross profitability, and return on assets.

To measure investors' attention, we construct a composite attention index based on six individual attention proxies, which include: (1) abnormal Google search volume, measured as the log difference between Google search volume for a stock in the past one month and the average search volume over the past year, following Da, Engelberg, and Gao (2011). The sample period is from July 2004 to December 2016; (2) media coverage, defined as the number of news articles published on Thomson Reuters News Analytics during the month for each stock. The news data sample period is from January 1996 to December 2021; (3) abnormal EDGAR download, measured as the log difference between the number of EDGAR downloads in the current month and the average number over the past year. ¹⁰ The sample period for the search records is from December 2003 to June 2017; (4) analyst coverage, represented by the total number of estimates for the earnings forecasts for the current fiscal year. The data sample period is from March 1983 to December 2021; (5) price delay, the average delay with which a firm's stock price responds to market information (Hou and Moskowitz, 2005). 11 The larger the number, the less attention the corresponding stock receives. Thus, we employ the reciprocal of the delay measure as the attention measure when constructing the

¹⁰The EDGAR downloads data are available at Ryans' website (http://www.jamesryans.com/) and we follow Ryans (2017)'s method to extract the human downloads.

¹¹We use the first delay measure in Hou, et al. (2005), which is $1 - \frac{R_{\delta_j}^2 n_{=0,\forall n \in [1,4]}}{R^2}$, where the denominator is R^2 from regression $r_{j,t} = \alpha_j + \beta_j R_{M,t} + \sum_{n=1}^4 \delta_j^{-n} R_{M,t-n} + \epsilon_{j,t}$, and the numerator is R^2 from the same regression by restricting $\delta_j^{-n} = 0$, $\forall n \in [1,4]$. In the regression, $r_{j,t}$ is the return on stock j and $R_{M,t}$ is the market return in month t calculated by Fama and French (1993).

composite attention index. The price delay data is from January 1980 to December 2021; (6) abnormal Bloomberg download, proposed by Ben-Rephael et al., (2017). Specifically, for each stock on each day, the abnormal Bloomberg download is a dummy variable that takes a value of one if Bloomberg's daily maximum score is 3 or 4 and zero otherwise. Then, for each stock in every month, we sum up the abnormal daily Bloomberg download indicators within the month. The data is obtained on Bloomberg terminals from February 2010 to December 2016. For each month, we first convert individual attention measures into ranks and standardize to obtain the z-score. Then, we construct the composite attention index as the average z-score of these six attention proxies.

To measure investors' disagreement, we construct a composite disagreement index based on three individual disagreement proxies, which include (1) stock return volatility, measured as the standard deviation of weekly stock excess returns (from Thursday to Wednesday) over the year ending on the portfolio formation date. The sample period is from January 1980 to December 2021; (2) analysts' return forecast dispersion, which is the standard deviation of analysts' return forecasts. Return forecasts are the 12-monthahead analysts' target prices forecasted in month t divided by the actual stock price at the beginning of month t. The sample period is from March 1999 to December 2021; (3) analysts' earnings forecast dispersion, which is the standard deviation of analysts' earnings forecast divided by the actual stock price in the same month. The sample period is from March 1982 to December 2021. We follow Han et al. (2022) to include both analysts' return, and earnings forecast dispersion when constructing the composite disagreement proxy. For each month, we first convert individual disagreement measures into ranks and standardize to obtain the z-score. Then, we construct the composite disagreement index as the average z-score of these three disagreement proxies.

Table 1 reports the summary statistics (Panel A) and the time-series average of the cross-sectional correlation coefficients (Panel B) between the variables used in the paper. Notably, the correlation coefficients between TO and Attention, as well as Disagreement are positive and moderately high, suggesting that TO may contain information about both investors' attention and disagreement. The correlation coefficient between MISP and Disagreement is the highest among all the variables,

indicating that MISP may be driven by investors' disagreement, which is consistent with the literature (Han et al., 2022). Panel C of Table 1 reports the monthly value-weighted portfolio average excess returns, CAPM alphas, and Fama-French three-factor alphas sorted by MISP, PEAD and FIN from 1980 to 2021. For example, the portfolio sorted by PEAD has an excess return spread of 0.49%, a CAPM alpha spread of 0.55%, and a Fama-French three-factor alpha spread of 0.60% per month. All of these are statistically significant.

[Insert Table 1 here]

4. Empirical Results

In this section, we report our main empirical results: trading volume mitigates PEAD and amplifies FIN. Specifically, volume mitigates PEAD in low-volume state and amplifies PEAD in high-volume state, and the mitigation effect dominates the amplification effect overall. Meanwhile, the volume amplification effect on FIN persists for stocks across different volume levels.

4.1 Volume mitigation effect on PEAD

We conduct an independent five-by-five double sort on PEAD and trading volume (TO) and obtain 25 value-weighted TO-PEAD portfolios, as shown in Table 2. Specifically, at the end of each month, we sort the stocks into five groups based on TO and sort the stocks into five groups based on PEAD independently. Finally, we have the 25 value-weighted TO-PEAD portfolios.

Panel A of Table 2 reports the value-weighted excess returns of the 25 portfolios, as well as the differences in excess returns for stocks among volume quintile 1, quintile 3 and quintile 5 portfolios. The spread of the PEAD L-S portfolio decreases along with the trading volume, especially among stocks in low-volume state. The PEAD L-S portfolio's average return decreases from 0.98% for volume quintile 1 portfolio to 0.60% for volume quintile 5 portfolio, with a difference of 0.37% (t-value = -1.62). Although insignificant, we do observe the volume mitigation effect on PEAD.

However, when we divide the stocks into low-volume state (quintile 1 to 3) and high-volume state (quintile 3 to 5), we observe a volume mitigation and amplification effect

in the two subsamples, respectively. Specifically, the average return of the L-S portfolio decreases by nearly 0.94% (t-value = -4.17) from volume quintile 1 portfolio to volume quintile 3 portfolio, while the returns increase 0.57% (t-value = 2.68) from volume quintile 3 portfolio to volume quintile 5 portfolio. Therefore, volume mitigates PEAD among stocks in low-volume state and amplifies PEAD among stocks in high-volume state, and the mitigation effect dominates the amplification effect, resulting in an overall mitigation effect.

[Insert Table 2 here]

To examine whether such result is driven by risks, we report the results for valueweighted CAPM-alphas and FF3 alphas of the 25 TO-PEAD portfolios in Panel B and Panel C of Table 2, respectively. The CAPM alpha of PEAD L-S portfolio decreases significantly from 0.99% for quintile 1 to 0.10% for quintile 3 portfolio, with a difference of -0.89% (t-value = -3.81), and then increases to 0.69% for quintile 5 portfolio, with a significant difference of 0.59% (t-value = 2.77). The total decrease of PEAD L-S portfolio CAPM alpha from volume quintile 1 to 5 portfolio is 0.30% (tvalue = -1.28). The FF3 alpha of PEAD L-S portfolio decreases from 0.94% for quintile 1 to 0.12% for quintile 3 portfolio, and then increases to 0.77% for quintile 5 portfolio, with differences of -0.83% (t-value = -3.39) and 0.65% (t-value = 3.00), respectively. The total decrease of PEAD L-S portfolio FF3 alpha from volume quintile 1 to 5 portfolio is 0.18% (t-value = -0.80). These results are consistent with those obtained from excess returns, which indicate that trading volume mitigates the mispricing captured by PEAD among stocks in low-volume state and amplifies such mispricing among stocks in high-volume state. Furthermore, the mitigation effect generally dominates the amplification effect, leading to an overall weak mitigation effect. 12

¹²When we use the equal-weighted method to calculate the alphas of portfolios (see Appendix Table A2 Panel A), we observe the mitigation effects among stocks in both low-volume state and high-volume state. And the magnitudes both decreased compared to the value-weighted results in the baseline. This is related to the fact that firm size increases from TO quintile 1 to 3 and then decrease from TO quintile 3 to 5. Size is positively related to investor attention but negatively related to disagreement. Hence, from TO quintile 1 to 3, the increase in firm size makes the disagreement related amplification effect weaker while attention related mitigation effect stronger that leads to a stronger mitigation effect among stocks in low-volume state when switch from equal- to value-weighted. Similarly, from TO quintile 3 to 5, the decrease in firm size makes the disagreement related amplification effect stronger while attention related mitigation effect weaker that leads to a stronger amplification effect among stocks in high-volume state when switch from equal- to value-weighted method. We further apply the independently triple sort on PEAD, TO and size to control the size effect (see Appendix Table A4 Panel A), consistently, both the magnitudes of mitigation effect among stocks in low-volume state and amplification effect among stocks in high-volume state decrease though still significant.

4.2 Volume amplification effect on FIN

In Table 3, we conduct an independent three-by-five double sort on FIN and trading volume. Specifically, at the end of each month, we sort the stocks into five groups based on TO and we follow Daniel, Hirshleifer, and Sun (2020) to sort the firms into three groups based on the 1-year NSI and 5-year CSI. We finally obtained the 15 value-weighted TO-FIN portfolios. Panel A of Table 3 reports the value-weighted excess returns of the 15 TO-FIN portfolios, as well as the differences in excess returns among volume quintile 1/3/5 portfolios. The spread of FIN long-short (L-S) portfolio increases monotonically in trading volume overall. The average return of L-S portfolios increases from 0.14% for volume quintile 1 portfolio to 0.52% for volume quintile 3 portfolio. It further increases to 0.74% for volume quintile 5 portfolio, showing a persistent volume amplification effect on FIN. The difference of L-S portfolios average return between volume quintile 5 and volume quintile 1 portfolios is 0.60% (t-value = 1.63).

[Insert Table 3 here]

Panel B and Panel C of Table 3 report the value-weighted CAPM alphas and FF3 alphas of the 15 TO-FIN portfolios, respectively. Consistent with the result of excess returns, the FIN L-S CAPM (FF3) alpha increases from 0.26% (0.30%) for volume quintile 1 portfolio to 0.63% (0.56%) for volume quintile 3 portfolio, and further increases to 1.04% (0.83%) for volume quintile 5 portfolio. The total difference of FIN L-S CAPM (FF3) alpha between volume quintile 5 and quintile 1 portfolio is 0.77% (0.53%) with a t-value of 2.02 (1.76).

4.3 Volume amplification effect on MISP

Han et al. (2022) conduct a five-by-five double sort on MISP and trading volume and find that volume amplifies MISP and this amplification effect is mainly concentrated among stocks in high-volume state. In Table 4, we conduct the same test and get consistent results. Panel A of Table 4 reports the value-weighted excess returns of the 25 portfolios, as well as the differences in average returns among volume quintile 1/3/5 portfolios. The spread of MISP long-short (L-S) portfolio increases along with the trading volume especially among stocks in high-volume state. The L-S portfolio return increases from 0.52% for volume quintile 1 portfolio to 1.36% for volume quintile 5

portfolio, with a difference of 0.84% (t-value=3.33). In particular, the L-S portfolio average return almost does not change (the difference is only 0.01% with t-value = 0.04) among stocks in low-volume state. In contrast, the average return of the L-S portfolio increases by 0.83% (t-value = 3.56) from quintile 3 portfolio to quintile 5 portfolio, accounting for the main increase from quintile 1 to 5 portfolio.

[Insert Table 4 here]

Such result is also consistent with our findings of the volume mitigation effect on PEAD and the volume amplification effect on FIN, because that MISP contains price momentum that is related to earnings momentum of PEAD (Chordia and Shivakumar, 2006). In addition, MISP incorporates the net stock issuance (NSI) and composite stock issuance (CSI), which are used to construct FIN. Therefore, based on the construction, MISP contains both types of mispricing that are captured by PEAD and FIN. Among stocks in low-volume state, the mitigation effect on PEAD and the amplification effect on FIN offset each other, and this is consistent with the MISP L-S return difference of 0.01% (t-value = 0.04) between quintile 1 and quintile 3 portfolio. Among stocks in high-volume state, the amplification effects on PEAD and FIN are enhanced, and this is consistent with the MISP L-S return difference of 0.83 (t-value = 3.56) between quintile 3 and quintile 5 portfolio.

To control for more potential risk factors, Panel B and Panel C of Table 4 report the CAPM adjusted alphas and FF3 adjusted alphas of the 25 TO-MISP portfolios, respectively. Consistent with the results for average returns, MISP L-S portfolio CAPM (FF3) alpha does not change too much from volume quintile 1 to quintile 3 portfolio, with a difference of 0.03% (0.12%) and t-value of 0.05 (-0.24). While the MISP L-S portfolio CAPM (FF3) alpha increases significantly from 0.72% (0.74%) for the quintile 3 portfolio to 1.72% (1.70%) for quintile 5 portfolio, with a difference of 1.00% (0.96%) and t-value of 4.48 (4.25).

¹³In Appendix Table A1, we show the average returns/CAPM alphas/FF3 alphas of the 25 portfolios double sorted on price momentum (MOM) and TO, where the MOM is the cumulative return over the past 12 months. For example, in Table A1 panel A, the spread of MOM L-S portfolio decrease among low-volume state stocks (although insignificant) and increase among high-volume state stocks, which display the similar pattern to PEAD.

Hou, Peng and Xiong (2009) document that trading volume, as a proxy of attention, displays a mitigation effect on PEAD for both low- and high-volume state stocks. However, we find a mitigation effect on PEAD for low-volume state stocks but an amplification effect for high-volume state stocks. This difference can be due to the different methods used. Hou, Peng and Xiong (2009) use the equal-weighted method, while we use the value-weighted method. Under the equal-weighted method, small firms play a larger role than under the value-weighted method.

For the PEAD anomaly, the underreaction to the earnings news matters more under the equal-weighted method and the mitigation effect may dominate the amplification effect even for the stocks in high-volume state. Hence, we can have a net mitigation effect on PEAD even for stocks in high-volume state under the equal-weighted method as documented in Hou, Peng and Xiong (2009) but a net amplification effect for stocks in high-volume state under the value-weighted method as documented by this study.

Indeed, when applying the equal-weighted method to calculate the portfolio returns, we find similar results to the literature. In Panel A of Table A2 in the Appendix, we report the equal-weighted CAPM adjusted alphas of the 25 TO-PEAD portfolios and FF3 adjusted alphas of PEAD L-S portfolios. The results are consistent with Hou, Peng and Xiong (2009)'s findings that L-S portfolio average returns and FF3 adjusted alphas decrease across all levels of trading volume.

As for MOM, it should be closely related to PEAD (Chordia and Shivakumar, 2006). Hence, under the equal-weighted method, the mitigation effect for MOM should be stronger than under the value-weighted method. Indeed, we find this is the case. The amplification effect (net the mitigation effect) in Panel B of Table A2 in the Appendix is weaker (stronger) than that in Panel C of Table A1.¹⁴

5. Economic Interpretation

In this section, we turn to interpret our main empirical results and investigate the underlying two common behavioral forces that drive mispricing. One is the investor's

¹⁴Although Hou, Peng and Xiong (2009) find the significant amplification effect of trading volume on MOM from low-volume groups to high-volume groups, the difference could be due to different sample selections. We also find a very strong amplification effect on MOM if restricted to their sample period (result upon request).

inattention to new information, which can be mitigated by improved attention. The other one is expectation bias (e.g., overconfidence), which can be amplified by an increase in disagreement. Therefore, in this section, we first aim to inspect the relation between trading volume and investors' attention or disagreement, respectively.

Since both behavioral forces could affect PEAD and they themselves are affected by volume via either the attention channel or the disagreement channel, we examine how trading volume affects PEAD via these two channels. Besides PEAD, we further examine the underlying channel through which trading volume affects other anomalies such as FIN, which is less affected by investor inattention but more by expectation bias, and MISP, which is a composite mispricing score that potentially captures the two types of mispricing and can be affected by both limited attention and expectation bias.

5.1 Investor attention

In this subsection, we examine the relation between investors' attention and trading volume. Although trading volume has been widely used to proxy investors' attention in many literatures (e.g., Barber and Odean, 2008; Chen, Tang, Yao, and Zhou, 2022; Chen, He, Tao, and Yu, 2022), it is considered to be the outcome of investors' attention. Thus, we propose some economic attention inputs to measure investors' attention and then test the relation between investors' attention and trading volume.

Panel A of Table 5 reports the value-weighted investors' attention of portfolios sorted by trading volume (TO). We use the average z-score of six individual attention proxies, including abnormal Google search volume, media coverage, abnormal EDGAR download, analyst coverage, price delay, and abnormal Bloomberg downloads to construct the composite attention index. In general, the investors' attention increases along with the trading volume, especially from volume quintile 1 portfolio to quintile 3 portfolio with a difference of 1.10 (t-value = 14.36). ¹⁵

Panel B of Table 5 reports the value-weighted investors' attention of portfolios double-sorted by PEAD and trading volume (TO). Specifically, among high PEAD portfolios, investors' attention increases from -0.14 for volume quintile 1 portfolio to 1.04 for the

¹⁵The value-weighted attention decreases slightly from TO quintile 3 to 5 which is consistent with the fact that the firm size also decreases from TO quintile 3 to 5. Thus, it generates the weaker mitigate effect on PEAD among low-volume state stocks when switch equal- to value-weighted method.

quintile 3 portfolio, with a difference of 1.18 (t-value = 22.74), and further increases to 1.07 for volume quintile 5 portfolio, with a difference of 0.04 (t-value = 1.09). The result indicates that trading volume acts as a proxy of investor attention among stocks in low-volume state. The lower the trading volume, the lower the investor attention, and the higher the mispricing captured by PEAD. This is consistent with our empirical result of the volume mitigation effect on PEAD among stocks in low-volume state.

Panel C and Panel D of Table 5 report the value-weighted investors' attention of 15 TO-FIN portfolios and 25 TO-MISP portfolios, respectively. They display a similar pattern to Panel A, where the attention increases with trading volume, especially for stocks in low-volume state.

[Insert Table 5 here]

5.2 Investor disagreement

In this subsection, we examine the relation between investors' disagreement and trading volume. Panel A of Table 6 reports the value-weighted investors' disagreement of portfolios sorted by trading volume (TO). We use the average z-score of three individual disagreement proxies, including stock return volatility, analysts' return forecast dispersion, and analysts' earnings forecast dispersion. In general, investor disagreement increases monotonically from volume quintile 1 to quintile 5 portfolios, which helps explain the volume amplification effect among stocks in high-volume state.

Panel B of Table 6 reports the value-weighted investors' disagreement of portfolios double-sorted by PEAD and trading volume (TO). Within each level of PEAD portfolios, investor disagreement increases monotonically from volume quintile 3 to quintile 5 portfolio, which helps explain the volume amplification effect among stocks in high-volume state. In contrast, among stocks in low-volume state, investor disagreement decreases from quintile 1 to quintile 3 portfolios for majority of PEAD portfolios, indicating that the mitigation effect could be even stronger than the level we observe in Table 2.

Panel C of Table 6 reports the value-weighted investors' disagreement of portfolios double-sorted by FIN and trading volume (TO). In general, for each FIN portfolio, the

investors' disagreement increases along with the trading volume. For instance, investors' disagreement for low-FIN portfolio increases by 0.77 (t-value = 20.16) from volume quintile 1 portfolio to quintile 5 portfolio. Furthermore, the disagreement increases mostly come from stocks in a high-volume state (i.e., from quintile 3 to quintile 5). Within each level of FIN portfolios, investors' disagreement monotonically increases from volume quintile 3 portfolio to quintile 5 portfolio. Similarly, as shown in Panel D of Table 6, disagreement increases in trading volume for double-sorted TO-MISP portfolios.

[Insert Table 6 here]

5.3 Fama-MacBeth analysis

In this section, we provide direct evidence about the amplification and mitigation effects of trading volume on different types of mispricing. Through investigating the relation between trading volume and investors' attention, as well as investors' disagreement, we find that investors' attention increases with trading volume, especially in low-volume state stocks. On the other hand, investors' disagreement increases with trading volume, especially in high-volume state stocks. Based on these findings, we conjecture that trading volume is more likely to proxy investors' attention among low-volume state stocks, while it is more likely to proxy investors' disagreement among high-volume state stocks. We test this conjecture through a series of Fama-MacBeth regressions including trading volume, factors that capture mispricing, investors' attention, disagreement, and their interaction with anomaly variables.

5.3.1 Volume mitigation effect on PEAD

Table 7 reports the Fama-Macbeth results for PEAD. The dependent variable is the one-month-ahead monthly excess return. The independent variables include PEAD, trading volume (TO), Attention and Disagreement. All the independent variables are converted into quintile ranks. We also consider the intersection terms: PEAD*TO, PEAD*Attention, and PEAD*Disagreement. We control firm size which is the logarithm of market capitalization and the logarithm of book-to-market ratio in all the regressions.

Panel A of Table 7 reports the Fama-Macbeth regressions results for the full sample. In

the first four regressions, we include only one of PEAD, trading volume (TO), Attention, and Disagreement as the main independent variable. Higher PEAD indicates greater undervaluation, suggesting that PEAD positively predicts future stock returns. The significant positive coefficients on PEAD (0.225 with a t-value of 13.32) in first regression is consistent with the theory. In the second regression, the coefficient on TO is negative but insignificant, which is consistent with prior studies that the relation between trading volume and future stocks return is ambiguous. In the third regression, the coefficient on Attention is positive and significant. Although we incorporate the retail attention (i.e., Google Search Volume) in our composite attention measure, the other attention measures are more likely to capture the institutional investors' attention (e.g., Edgar download, Bloomberg download and analyst coverage). Thus, the positive coefficient on Attention is consistent with the existing literature that institutional investors' attention is usually positively related to future stock return (Da et al., 2022). The coefficient on investors' disagreement is negative in regression (4) which is also consistent with the theory that stocks with higher disagreement are usually overvalued and negatively related to future returns. Furthermore, the coefficients of PEAD*TO, PEAD*Attention and PEAD*disagreement are negative, negative, and positive in regression (5), (6) and (7), respectively, which are also consistent with the economic rationales that TO (as a proxy for Attention) and Attention could reduce mispricing due to limited attention, while Disagreement can amplify the mispricing due to investor's bias.

[Insert Table 7 here]

We focus on the coefficients on the interaction terms of PEAD and trading volume, investors' attention, as well as disagreement. The results based on the portfolio analysis predict that the coefficients of PEAD*TO should be negative since trading volume can help mitigate PEAD. After attention and its interaction term with PEAD are added into the regression, the magnitude of negative coefficients on PEAD*TO should reduce. Regressions (5) and (8) in Panel A of Table 7 confirm this prediction. We can see that the interaction term of PEAD*TO is -0.0397 (t-value = -4.70) and its magnitude decreases to -0.0287 (t-value = -3.29) when we control the attention variables. It indicates that trading volume is more like an attention proxy for PEAD. Moreover, the magnitude of coefficient on PEAD*TO becomes larger (-0.0517) in regression (9) when

we include disagreement variables. The reason is that after controlling for disagreement, trading volume becomes a purer proxy for attention which would further mitigate the mispricing. In regression (10), when we include all the variables as the independent variables, the magnitude of coefficient on PEAD*TO decreases relative to regression (9), which further indicates that trading volume is more likely to measure attention for PEAD.

We further divide the sample into low-volume state and high-volume state subsample based on trading volume quintile ranks. The low-volume state subsample are stocks from trading volume quintile 1 to quintile 3 and the high-volume subsample are stocks from trading volume quintile 3 to quintile 5.

Panel B of Table 7 is our focus for PEAD, and it reports the results of low-volume state subsample. The results based on the portfolio analysis indicate that the coefficients of PEAD*TO in low-volume state subsample should be more negative and significant, and the magnitude would also decrease when attention and its interaction term with PEAD are added. Regressions (11) and (12) in Panel B of Table 7 confirm this prediction. We can see that the interaction term of PEAD*TO is -0.0478 (t-value = -3.24) of which the magnitude is larger than its coefficient in full sample (see regression (5)). The magnitude also decreases when we add attention variables in regression (12) which is -0.0323 (t-value = -2.05). In regression (14), we include all the variables as the independent variables. As expected, the magnitude of coefficient on PEAD*TO becomes smaller relatively to regression (13) and it is similar to regression (11), which confirms that trading volume is more likely to measure attention for PEAD especially among stocks in low-volume state.

Panel C of Table 7 reports the results of high-volume state subsample. Consistent with our conjecture, the coefficient of PEAD*TO becomes insignificant in high-volume state stocks (see regressions (15) and (16)). But it becomes significantly negative when we add disagreement variables in regression (17), which is consistent with the previous findings that TO becomes a purer attention measure after controlling disagreement. Furthermore, the magnitude of coefficient on PEAD*TO becomes smaller in regression

Overall, the Fama-MacBeth analysis in Table 7 confirms our portfolio analysis in prior sections that the trading volume could mitigate the PEAD because it is more likely to represent attention information for anomalies driven by limited attention and such effect is more pronounced in low-volume state stocks.

5.3.2 Volume amplification effect on FIN

We employ similar Fama-MacBeth regressions for anomaly FIN and report the empirical results in Table 8. The independent variables include FIN, trading volume (TO), Attention, and Disagreement. Consistent with Daniel, Hirshleifer and Sun (2020), we convert FIN into tertile ranks and all the other variables are converted into quintile ranks. Panel A of Table 8 reports the Fama-Macbeth regressions results for the full sample. Higher FIN indicates greater overvaluation, and thus the coefficient on FIN in regression (1) is significantly negative. Furthermore, the coefficients of FIN*TO, FIN*Attention and FIN*disagreement are negative, insignificant, and negative in regression (5), (6) and (7), respectively, which are also consistent with the economic rationales that TO (as a proxy for Disagreement) and Disagreement could amplify FIN mispricing, while Attention is unrelated to FIN mispricing.¹⁷

[Insert Table 8 here]

We focus on the coefficients on the interaction terms of FIN and trading volume, investors' attention, and disagreement. The coefficient of FIN*TO is negative and significant in regression (5) and it becomes insignificant when we add disagreement variables in regression (8). Such findings confirm the portfolio analysis results that trading volume is more likely to represent the disagreement information for mispricing driven by investors' bias.

¹⁶Although we do not observe the significant amplification effect of TO on PEAD mispricing among high-volume state stocks using Fama-Macbeth method, the insignificant result is consistent with the double sort results using equal-weighted method.

¹⁷If we add the triple interaction term FIN*Attention*Disagreement in the regression, the coefficient on the triple interaction term is significant and negative which indicates the attention could enlarge the amplification effect of disagreement on FIN mispricing.

We further divide the sample into low-volume state and high-volume state subsample based on its quintile ranks. Panel B of Table 8 reports the results of low-volume state subsample. Consistent with our conjecture, the coefficients of FIN*TO all become insignificant in low-volume state stocks (see regressions (11) to (14)).

Panel C of Table 8 reports the results of high-volume state subsample. The magnitude of coefficient on FIN*TO in high-volume state groups is larger than that in full sample (see regressions (15) and (5)). And the coefficient of FIN*TO becomes insignificant when we add disagreement and its interaction term with TO in regression (17). These findings are consistent with portfolio analysis results that the trading volume is more likely to measure disagreement for FIN especially in high-volume state stocks.

Overall, the Fama-MacBeth analysis in Table 8 confirms our portfolio analysis in prior sections that trading volume could amplify the FIN because it is more likely to measure disagreement information for anomalies driven by investors' bias and such effect is more pronounced in high-volume state stocks.

5.3.3 Volume amplification effect on MISP

We further employ the similar Fama-MacBeth regressions for anomaly MISP and Table 9 reports empirical results. Panel A of Table 9 reports the Fama-Macbeth regressions results of full sample. In the first four regressions, we only include one of MISP, trading volume (TO), Attention, and Disagreement as the main independent variable. In the first four regressions, we only include one of MISP, trading volume (TO), Attention, and Disagreement as the main independent variable. Higher MISP indicates greater overvaluation, suggesting that MISP negatively predicts future stock returns. The negative coefficient on MISP (-0.326 with a t-value of -8.49) in first regression is consistent with the theory. The coefficients of MISP*TO, MISP*Attention and MISP*disagreement are negative, positive and negative in regression (5), (6) and (7), respectively. Since MISP is related to both limited attention and investor bias, TO and Disagreement could amplify mispricing of MISP while Attention could reduce mispricing of MISP.

The regression coefficients on the interaction terms of MISP and trading volume, investors' attention and disagreement are the focus in this paper. The results based on the portfolio analysis predict that the coefficients of MISP*TO should be negative, and when disagreement and its interaction term with MISP are added into the regression, the magnitude of negative coefficients on MISP*TO should reduce. The regressions (5) and (9) in Panel A of Table 9 confirm this prediction. We can see that the interaction term of MISP*TO is -0.0340 (t-value = -2.76) and it becomes insignificant when we add disagreement variables. It indicates trading volume is more like a disagreement proxy for MISP. Moreover, the magnitude of coefficient on MISP*TO becomes larger (-0.0533) and even more significant (t-value = -4.17) in regression (8) where we include attention variables. The reason is that after controlling for attention, trading volume becomes a purer proxy for disagreement which would further amplify the mispricing. In regression (10), we include all the variables as the independent variables, and the coefficients of MISP*TO becomes insignificant, which further confirms that trading volume is more likely to measure disagreement for MISP.

We further divide the sample into low-volume state and high-volume state subsample based on trading volume quintile ranks. The low-volume state subsample consists of stocks from trading volume quintile 1 to quintile 3 and high-volume state subsample consists of stocks from trading volume quintile 3 to quintile 5.

Panel B of Table 9 reports the results of low-volume state subsample. Consistent with our conjecture, the coefficient of MISP*TO becomes insignificant in low-volume state stocks (see regression (11)) and it even becomes positive (see regressions (13) and (14)) when we include disagreement variables because trading volume is no longer a disagreement proxy for low-volume state stocks, and it may contain other information which could mitigate the MISP instead.

Panel C of Table 9 reports the results of high-volume state subsample. The results based on the portfolio analysis indicate that the coefficients of MISP*TO in high-volume state subsample should be more negative and significant, and the magnitude would also decrease when disagreement and its interaction term with MISP added. The regressions (15) in Panel C of Table 9 confirm this prediction. We can see that the interaction term

of MISP*TO is -0.103 (t-value = -5.19) of which the magnitude is larger than its coefficient in full sample (see regression (5)). The magnitude also decreases when we add disagreement and its interaction term with TO in regression (17) which is -0.0473 (t-value = -2.56), which confirms that trading volume is more likely to measure disagreement for MISP especially in high-volume state stocks.

Overall, the Fama-MacBeth analysis in Table 9 confirms our portfolio analysis in prior sections that trading volume could amplify the MISP because it is more likely to measure disagreement information for anomalies driven by investors' bias and such effect is more pronounced in high-volume state stocks.

5.4 Alternative explanations

In the previous sections, we demonstrated that trading volume is a sensible measure of investor attention among stocks in low-volume state and represents investor disagreement among stocks in high-volume state. However, various alternative explanations have been proposed in the literature, suggesting that trading volume may convey information beyond investor attention and disagreement. In this section, we explore three additional explanations that could potentially lead to a varied volume-return relationship, including arbitrage costs, illiquidity, and private information. We then show that our findings and interpretations remain consistent even after accounting for these factors.

Regarding the first explanation that trading volume might be indicative of arbitrage costs, Stambaugh, Yu, and Yuan (2015) discovered a negative IVOL-return relationship for overpriced stocks and a positive one for underpriced stocks, interpreting IVOL as a proxy for arbitrage costs. Given the positive correlation between trading volume and IVOL, as well as the similarity between IVOL-return and volume-return relationships, it is plausible to suggest that trading volume serves as a proxy for arbitrage costs, akin to IVOL. Consequently, we examine whether the volume mitigation and amplification effect persist after accounting for IVOL. We calculate IVOL as the standard deviation of residuals from Carhart four-factor regressions, using the previous month's daily returns following Ang et al. (2006).

For the second explanation, high trading volume indicates greater liquidity, while low trading volume implies greater illiquidity. Consequently, prior research, such as Glosten and Harris (1988) and Brennan and Subrahmanyam (1995), demonstrates that trading volume is a crucial factor in determining market liquidity. To examine if the volume-return relationship is linked to illiquidity, we employ the bid-ask spread as an illiquidity measure. A higher bid-ask spread corresponds to increased illiquidity, and we utilize Corwin and Schultz (2011)'s approach to calculate the bid-ask spread.

As for the final explanation, trading volume captures information asymmetry or private information. Easley, Kiefer, O'Hara, and Paperman (1996) contend that the likelihood of information-based trading is lower for high-volume stocks, suggesting that trading volume reflects the extent of private information. We adopt Easley, Prado, and O'Hara (2012)'s method to construct the volume-synchronized probability of informed trading (VPIN) to verify whether our primary findings persist after accounting for private information.

Thus, in line with Han et al. (2022), we employ IVOL, bid-ask spread from Corwin and Schultz (2011), and the volume-synchronized probability of informed trading (VPIN) from Easley, Prado, and O'Hara (2012) to gauge arbitrage costs, illiquidity, and private information, respectively. Utilizing Fama-MacBeth regressions, we investigate their impacts on the volume mitigation and amplification effect, while controlling firm characteristics like size and book-to-market ratios.

Table 10 displays the results, with all independent variables converted into quintile ranks. Panel A presents the alternative explanation tests for the trading volume mitigation effect on PEAD. The first regression includes PEAD, trading volume, and their interaction, excluding other competing variables. The negative coefficient of the PEAD interaction with TO indicates that trading volume negatively predicts future stock returns for underpriced stocks, while positively predicting future stock returns for overpriced stocks. This aligns with the mitigation effect of trading volume on PEAD. Regression (2) accounts for the IVOL effect, regression (3) controls for the bid-ask spread, and regression (4) adjusts for VPIN. Regression (5) includes all three alternative explanations. The consistently negative and significant coefficients of the PEAD interaction with TO across all regressions suggest that the trading volume's mitigation

effect on PEAD is robust to alternative explanations.

Panel B and Panel C present the alternative explanation tests for the trading volume amplification effect on FIN and MISP, respectively. For FIN, the coefficients of the FIN interaction with TO remain negative and significant across all regressions. As for MISP, the coefficients of the MISP interaction with TO persist as negative and significant across all regressions, except for the case of they become smaller but still significant at the 10% level when accounting for the IVOL effect. The results suggest that the trading volume's amplification effect on FIN and MISP is also robust to alternative explanations.

6. Conclusion

Our study highlights trading volume's dual function as both an indicator of disagreement and attention, the prominence of which fluctuates depending on the volume state. We find that the impact of trading volume on mispricing is determined by both the volume state and the root of the mispricing, whether it arises from investor bias or limited attention. Specifically, an increase in trading volume may mitigate mispricing resulting from limited attention when it reflects attention in states of low volume. However, when trading volume represents disagreement in high-volume states, it can amplify mispricing driven by investor bias, via exacerbating such bias.

These insights illuminate a nuanced interplay between trading volume, attention, disagreement, and mispricing, offering valuable knowledge into an area that has been relatively under-researched in current literature. This work advances our understanding of these multifaceted dynamics and may have significant applications for future studies.

Reference

Ali, Usman, and David Hirshleifer. "Shared analyst coverage: Unifying momentum spillover effects." Journal of Financial Economics 136, no. 3 (2020): 649-675.

Amromin, Gene, and Steven A. Sharpe. "From the horse's mouth: Economic conditions and investor expectations of risk and return." Management Science 60, no. 4 (2014): 845-866.

Ang, Andrew, Robert J. Hodrick, Yuhang Xing, and Xiaoyan Zhang. "The cross-section of volatility and expected returns." The Journal of Finance 61, no. 1 (2006): 259-299.

Atmaz, Adem, and Suleyman Basak. "Belief dispersion in the stock market." The Journal of Finance 73, no. 3 (2018): 1225-1279.

Bacchetta, Philippe, Elmar Mertens, and Eric Van Wincoop. "Predictability in financial markets: What do survey expectations tell us?." Journal of International Money and Finance 28, no. 3 (2009): 406-426.

Banerjee, Snehal. "Learning from prices and the dispersion in beliefs." The Review of Financial Studies 24, no. 9 (2011): 3025-3068.

Barber, Brad M., and Terrance Odean. "All that glitters: The effect of attention and news on the buying behavior of individual and institutional investors." The Review of Financial Studies 21, no. 2 (2008): 785-818.

Barberis, Nicholas. "Psychology-based models of asset prices and trading volume." Handbook of Behavioral Economics: applications and foundations 1, vol. 1, pp. 79-175. North-Holland, 2018.

Barberis, Nicholas, Andrei Shleifer, and Robert Vishny. "A model of investor sentiment." Journal of Financial Economics 49.3 (1998): 307-343.

Ben-Rephael, Azi, Zhi Da, and Ryan D. Israelsen. "It depends on where you search: Institutional investor attention and underreaction to news." The Review of Financial Studies 30, no. 9 (2017): 3009-3047.

Boehme, Rodney D., Bartley R. Danielsen, and Sorin M. Sorescu. "Short-sale constraints, differences of opinion, and overvaluation." Journal of Financial and Quantitative Analysis 41, no. 2 (2006): 455-487.

Brennan, Michael J., and Avanidhar Subrahmanyam. "Investment analysis and price formation in securities markets." Journal of Financial Economics 38, no. 3 (1995): 361-381.

Chen, Jian, Guohao Tang, Jiaquan Yao, and Guofu Zhou. "Investor attention and stock returns." Journal of Financial and Quantitative Analysis 57, no. 2 (2022): 455-484.

Chen, Xin, Wei He, Libin Tao, and Jianfeng Yu. "Attention and Underreaction-Related Anomalies." Management Science (2022).

Chordia, Tarun, Avanidhar Subrahmanyam, and V. Ravi Anshuman. "Trading activity and expected stock returns." Journal of Financial Economics 59.1 (2001): 3-32.

Chordia, Tarun, and Lakshmanan Shivakumar. "Earnings and price momentum." Journal of Financial Economics 80, no. 3 (2006): 627-656.

Chui, Andy CW, Sheridan Titman, and KC John Wei. "Individualism and momentum around the world." The Journal of Finance 65, no. 1 (2010): 361-392.

Cohen, Lauren, and Andrea Frazzini. "Economic links and predictable returns." The Journal of Finance 63, no. 4 (2008): 1977-2011.

Corwin, Shane A., and Paul Schultz. "A simple way to estimate bid-ask spreads from daily high and low prices." The Journal of Finance 67, no. 2 (2012): 719-760.

Da, Zhi, Jian Hua, Chih-Ching Hung, and Lin Peng. "Market returns and a tale of two types of attention." Available at SSRN 3551662 (2022).

Da, Zhi, Joseph Engelberg, and Pengjie Gao. "In search of attention." The Journal of Finance 66, no. 5 (2011): 1461-1499.

Da, Zhi, Umit G. Gurun, and Mitch Warachka. "Frog in the pan: Continuous information and momentum." The Review of Financial Studies 27, no. 7 (2014): 2171-2218.

Da, Zhi, Xing Huang, and Lawrence J. Jin. "Extrapolative beliefs in the cross-section: What can we learn from the crowds?." Journal of Financial Economics 140, no. 1 (2021): 175-196.

Daniel, Kent, David Hirshleifer, and Avanidhar Subrahmanyam. "Investor psychology and security market under-and overreactions." the Journal of Finance 53, no. 6 (1998): 1839-1885.

Daniel, Kent, David Hirshleifer, and Lin Sun. "Short-and long-horizon behavioral factors." The Review of Financial Studies 33, no. 4 (2020): 1673-1736.

Daniel, Kent, and Sheridan Titman. "Market reactions to tangible and intangible information." The Journal of Finance 61.4 (2006): 1605-1643.

DellaVigna, Stefano, and Joshua M. Pollet. "Investor inattention and Friday earnings announcements." The Journal of Finance 64, no. 2 (2009): 709-749.

Diether, Karl B., Christopher J. Malloy, and Anna Scherbina. "Differences of opinion and the cross section of stock returns." The Journal of Finance 57, no. 5 (2002): 2113-2141.

Easley, David, Marcos M. López de Prado, and Maureen O'Hara. "Flow toxicity and liquidity in a high-frequency world." The Review of Financial Studies 25, no. 5 (2012): 1457-1493.

Easley, David, Nicholas M. Kiefer, Maureen O'hara, and Joseph B. Paperman. "Liquidity, information, and infrequently traded stocks." The Journal of Finance 51, no.

4 (1996): 1405-1436.

Eisdorfer, Assaf, Kenneth Froot, Gideon Ozik, and Ronnie Sadka. "Competition links and stock returns." The Review of Financial Studies 35, no. 9 (2022): 4300-4340.

Eyster, Erik, Matthew Rabin, and Dimitri Vayanos. "Financial markets where traders neglect the informational content of prices." The Journal of Finance 74, no. 1 (2019): 371-399.

Gervais, Simon, Ron Kaniel, and Dan H. Mingelgrin. "The high-volume return premium." The Journal of Finance 56, no. 3 (2001): 877-919.

Glosten, Lawrence R., and Lawrence E. Harris. "Estimating the components of the bid/ask spread. "Journal of Financial Economics 21, no. 1 (1988): 123-142.

Fama, Eugene F., and Kenneth R. French. "Common risk factors in the returns on stocks and bonds." Journal of Financial Economics 33.1 (1993): 3-56.

Goetzmann, William N., and Massimo Massa. "Dispersion of opinion and stock returns." Journal of Financial Markets 8, no. 3 (2005): 324-349.

Greenwood, Robin, and Andrei Shleifer. "Expectations of returns and expected returns." The Review of Financial Studies 27, no. 3 (2014): 714-746.

Grinblatt, Mark, and Matti Keloharju. "Sensation seeking, overconfidence, and trading activity." The Journal of Finance 64, no. 2 (2009): 549-578.

Han, Yufeng, Dashan Huang, Dayong Huang, and Guofu Zhou. "Expected return, volume, and mispricing." Journal of Financial Economics 143, no. 3 (2022): 1295-1315.

Hirshleifer, David, Sonya Seongyeon Lim, and Siew Hong Teoh. "Driven to distraction: Extraneous events and underreaction to earnings news." The Journal of Finance 64, no. 5 (2009): 2289-2325.

Hoberg, Gerard, and Gordon Phillips. "Product market synergies and competition in mergers and acquisitions: A text-based analysis." The Review of Financial Studies 23, no. 10 (2010): 3773-3811.

Hoberg, Gerard, and Gordon Phillips. "Text-based network industries and endogenous product differentiation." Journal of Political Economy 124, no. 5 (2016): 1423-1465.

Hong, Harrison, and David A. Sraer. "Speculative betas." The Journal of Finance 71, no. 5 (2016): 2095-2144.

Hong, Harrison, and Jeremy C. Stein. "Disagreement and the stock market." Journal of Economic Perspectives 21, no. 2 (2007): 109-128.

Hong, Harrison, Terence Lim, and Jeremy C. Stein. "Bad news travels slowly: Size, analyst coverage, and the profitability of momentum strategies." The Journal of Finance 55, no. 1 (2000): 265-295.

Hou, Kewei, and Tobias J. Moskowitz. "Market frictions, price delay, and the cross-section of expected returns." The Review of Financial Studies 18, no. 3 (2005): 981-1020.

Hou, Kewei, Wei Xiong, and Lin Peng. "A tale of two anomalies: The implications of investor attention for price and earnings momentum." Available at SSRN 976394 (2009).

Israeli, Doron, Ron Kaniel, and Suhas A. Sridharan. "The real side of the high-volume return premium." Management Science 68, no. 2 (2022): 1426-1449.

Kaniel, Ron, Arzu Ozoguz, and Laura Starks. "The high-volume return premium: Cross-country evidence." Journal of Financial Economics 103, no. 2 (2012): 255-279.

Lee, Charles MC, and Bhaskaran Swaminathan. "Price momentum and trading volume." The Journal of Finance 55, no. 5 (2000): 2017-2069.

Liao, Jingchi, Cameron Peng, and Ning Zhu. "Extrapolative bubbles and trading volume." The Review of Financial Studies 35, no. 4 (2022): 1682-1722.

Llorente, Guillermo, Roni Michaely, Gideon Saar, and Jiang Wang. "Dynamic volume-return relation of individual stocks." The Review of Financial Studies 15, no. 4 (2002): 1005-1047.

Lo, Andrew W., and Jiang Wang. "Trading volume: definitions, data analysis, and implications of portfolio theory." The Review of Financial Studies 13, no. 2 (2000): 257-300.

Lou, Xiaoxia, and Tao Shu. "Price impact or trading volume: Why is the Amihud (2002) measure priced?." The Review of Financial Studies 30, no. 12 (2017): 4481-4520.

Medhat, Mamdouh, and Maik Schmeling. "Short-term momentum." The Review of Financial Studies 35, no. 3 (2022): 1480-1526.

Odean, Terrance. "Volume, volatility, price, and profit when all traders are above average." The journal of finance 53, no. 6 (1998): 1887-1934.

Parsons, Christopher A., Riccardo Sabbatucci, and Sheridan Titman. "Geographic leadlag effects." The Review of Financial Studies 33, no. 10 (2020): 4721-4770.

Ryans, James. "Using the EDGAR log file data set." Available at SSRN 2913612 (2017).

Stambaugh, Robert F., and Yu Yuan. "Mispricing factors." The Review of Financial Studies 30, no. 4 (2017): 1270-1315.

Table 1 Summary Statistics

This table reports summary statistics and correlation coefficients for trading volume, three main anomalies, composite attention index and composite disagreement index. Trading volume (TO) is the average turnover over the prior 12 months where the monthly turnover is the number of shares traded during a month divided by the number of shares outstanding at the end of the month. PEAD is the 4-day cumulative abnormal return around earnings announcement following Daniel, Hirshleifer and Sun (2020). FIN is based on the 1-year net share issuance (NSI) and 5-year composite share issuance (CSI). We follow Daniel, Hirshleifer and Sun (2020)'s method to sort firms into three financing groups (low "L", middle "M", or high "H") and take value of one for L, two for M and three for H groups. MISP is composite mispricing score which is the average rankings of 11 anomalies including net stock issues, composite equity issues, accruals, net operating assets, asset growth, investment to assets, financial distress, O-score, momentum, gross profitability, and return on assets from Stambaugh and Yuan (2017). Attention is the average z-score of six individual attention measures which include abnormal Google search volume, media coverage, abnormal EDGAR download, analyst coverage, price delay and abnormal Bloomberg download. Disagreement is the average z-score of three individual disagreement measures which include stock return volatility, analysts' return forecast dispersion and analysts' earnings forecast dispersion. Panel A reports the time-series average (mean), standard deviation (std), minimum (min), quantile 25 (Q25), median, quantile 75 (Q75) and maximum (max). Panel B reports the time-series average of the cross-sectional Pearson correlation between all the variables. Panel C reports anomalies' monthly average excess returns, CAPM alphas and FF3 alphas. The sample period is from January 1980 to December 2021. Value weighted monthly excess returns and alphas are reported in percentages and t-statistics are reported in parenthe

Panel A: Summar	y statistics							
variable	mean	std	min	Q25	median	Q75	max	
ТО	1.17	2.60	0.01	0.38	0.78	1.37	89.41	
PEAD	0.00	0.09	-0.59	-0.04	0.00	0.04	1.24	
FIN	2.09	0.48	1.00	2.00	2.00	2.00	3.00	
MISP	49.13	13.38	10.56	39.70	48.29	57.78	94.68	
Attention	0.00	1.00	-2.18	-0.77	-0.01	0.77	2.28	
Disagreement	0.00	1.00	-1.83	-0.82	-0.02	0.84	1.78	
Panel B: Correlati	ons							
variable	TO	PEAD	FIN	MISP	Attention	Disagreement		
ТО	1.00							
PEAD	-0.02	1.00						
FIN	0.16	-0.01	1.00					
MISP	0.11	-0.09	0.52	1.00				
Attention	0.24	-0.01	-0.02	-0.11	1.00			
Disagreement	0.27	0.00	0.21	0.33	-0.14	1.00		

	Average excess return				FF3 alpha				
	S	L	L-S	S	L	L-S	S	L	L-S
DEAD	0.49 *	0.98 ***	0.49 ***	-0.39 ***	0.15 **	0.55 ***	-0.40 ***	0.20 ***	0.60 ***
PEAD	(1.89)	(4.14)	(4.34)	(-5.76)	(2.20)	(5.05)	(-6.18)	(3.31)	(5.96)
	L	S	L-S	L	S	L-S	L	S	L-S
FIN	1.07 ***	0.47 *	0.60 ***	0.43 ***	-0.44 ***	0.87 ***	0.36 ***	-0.38 ***	0.73 ***
	(6.15)	(1.72)	(3.13)	(3.29)	(-4.74)	(4.29)	(3.89)	(-4.87)	(5.32)
	L	S	L-S	L	S	L-S	L	S	L-S
MISP	0.88 ***	0.04	0.84 ***	0.25 ***	-0.93 ***	1.17 ***	0.24 ***	-0.95 ***	1.19 ***
	(4.94)	(0.14)	(4.22)	(3.64)	(-7.16)	(6.36)	(4.22)	(-7.91)	(7.23)

Table 2 Average returns and alphas of portfolios sorted by PEAD and volume

This table reports the value-weighted average excess returns, CAPM alphas and FF3 alphas of portfolios double sorted on PEAD and trading volume (TO). Short leg refers to the quintile with the lowest PEAD and Long leg refers to the quintile with the highest PEAD. L-S (H-L) refers to the long-short PEAD (high-minus-low volume) portfolio spread. All portfolios are rebalanced monthly. Value-weighted monthly excess returns and alphas are reported in percentages and t-statistics are reported in parentheses. The sample period is from January 1980 to December 2021.

	Low	2	Medium Volume	4	High Volume	M-L	Н-М	H-L
	Volume							
Short	0.17	0.46 *	0.80 ***	0.43	0.44	0.63 ***	-0.36	0.27
	(0.72)	(1.93)	(3.59)	(1.60)	(1.28)	(3.84)	(-1.56)	(1.02)
2	0.61 ***	0.81 ***	0.81 ***	0.64 ***	0.85 ***	0.20	0.04	0.24
	(2.69)	(4.69)	(3.77)	(2.78)	(2.59)	(1.22)	(0.22)	(0.93)
3	0.81 ***	0.73 ***	0.80 ***	0.74 ***	0.81 **	-0.01	0.01	0.00
	(4.81)	(4.36)	(4.52)	(3.24)	(2.45)	(-0.10)	(0.06)	(0.00)
4	0.82 ***	0.77 ***	0.75 ***	0.77 ***	0.83 ***	-0.06	0.08	0.02
	(3.92)	(4.34)	(3.82)	(3.52)	(2.71)	(-0.43)	(0.39)	(0.07)
Long	1.14 ***	1.22 ***	0.84 ***	1.00 ***	1.04 ***	-0.31 *	0.20	-0.10
	(4.56)	(5.45)	(3.95)	(4.14)	(3.16)	(-1.69)	(0.84)	(-0.34)
L-S	0.98 ***	0.76 ***	0.04	0.56 ***	0.60 ***	-0.94 ***	0.57 ***	-0.37
	(6.19)	(4.29)	(0.24)	(3.70)	(3.66)	(-4.17)	(2.68)	(-1.62)

nel B: CAPN	M alpha							
Short	-0.46 ***	-0.24	0.07	-0.45 ***	-0.69 ***	0.53 ***	-0.76 ***	-0.22
	(-2.79)	(-1.62)	(0.57)	(-3.77)	(-4.37)	(3.20)	(-3.46)	(-0.86)
2	0.07	0.25 **	0.16 *	-0.15 *	-0.17	0.09	-0.33 *	-0.25
	(0.41)	(2.17)	(1.71)	(-1.82)	(-1.15)	(0.52)	(-1.90)	(-0.99)
3	0.35 **	0.20 *	0.19 *	-0.01	-0.24 *	-0.16	-0.43 **	-0.59 ***
	(2.52)	(1.78)	(1.75)	(-0.09)	(-1.77)	(-1.20)	(-2.31)	(-2.61)
4	0.30 *	0.22 **	0.10	0.01	-0.12	-0.21	-0.22	-0.42 *
	(1.92)	(2.02)	(1.07)	(0.09)	(-0.90)	(-1.36)	(-1.28)	(-1.95)
Long	0.53 ***	0.58 ***	0.17	0.18 *	0.00	-0.36 *	-0.17	-0.52 *
	(2.69)	(3.45)	(1.39)	(1.87)	(0.00)	(-1.92)	(-0.66)	(-1.68)
L-S	0.99 ***	0.82 ***	0.10	0.62 ***	0.69 ***	-0.89 ***	0.59 ***	-0.30
	(6.11)	(5.12)	(0.68)	(3.96)	(4.28)	(-3.81)	(2.77)	(-1.28)

el C: FF3 a	-							
Short	-0.54 ***	-0.32 **	0.03	-0.49 ***	-0.64 ***	0.56 ***	-0.67 ***	-0.10
	(-3.98)	(-2.42)	(0.21)	(-4.11)	(-4.80)	(3.42)	(-3.29)	(-0.50)
2	-0.04	0.22 *	0.12	-0.20 **	-0.13	0.16	-0.25	-0.09
	(-0.30)	(1.85)	(1.31)	(-2.55)	(-0.94)	(1.13)	(-1.46)	(-0.44)
3	0.27 **	0.14	0.13	-0.06	-0.18	-0.14	-0.31 *	-0.45 **
	(2.17)	(1.50)	(1.56)	(-0.60)	(-1.35)	(-0.99)	(-1.83)	(-2.11)
4	0.20 *	0.18	0.07	0.00	-0.04	-0.14	-0.11	-0.24
	(1.73)	(1.64)	(0.73)	(0.00)	(-0.33)	(-1.11)	(-0.64)	(-1.37)
Long	0.41 ***	0.53 ***	0.14	0.20 **	0.13	-0.26 *	-0.02	-0.28
	(3.02)	(3.51)	(1.19)	(2.15)	(0.97)	(-1.65)	(-0.09)	(-1.27)
L-S	0.94 ***	0.85 ***	0.12	0.69 ***	0.77 ***	-0.83 ***	0.65 ***	-0.18
	(5.78)	(5.29)	(0.79)	(4.46)	(4.88)	(-3.39)	(3.00)	(-0.80)

Table 3 Average returns and alphas of portfolios sorted by FIN and volume

This table reports the value-weighted average excess returns, CAPM alphas and FF3 alphas of portfolios double sorted on FIN and trading volume (TO). Long leg refers to the quintile with the lowest FIN and Short leg refers to the quintile with the highest FIN. L-S (H-L) refers to the long-short FIN (high-minus-low volume) portfolio spread. All portfolios are rebalanced monthly. Value-weighted monthly excess returns and alphas are reported in percentages and t-statistics are reported in parentheses. The sample period is from January 1980 to December 2021.

nel A: Avera	ge return							
	Low	2	Medium	4	High	M-L	H-M	H-L
	Volume	2	Volume	7	Volume	IVI-L	11-101	11-12
Long	0.95 ***	0.99 ***	1.01 ***	1.17 ***	1.33 ***	0.06	0.32	0.38
	(4.17)	(5.91)	(5.23)	(4.88)	(4.94)	(0.34)	(1.49)	(1.54)
2	0.74 ***	0.74 ***	0.79 ***	0.75 ***	0.92 ***	0.05	0.12	0.18
	(4.35)	(4.49)	(4.02)	(3.34)	(3.04)	(0.43)	(0.69)	(0.80)
Short	0.81 ***	0.54 **	0.50 **	0.33	0.59	-0.31	0.10	-0.21
	(2.93)	(2.40)	(2.25)	(1.13)	(1.59)	(-1.32)	(0.36)	(-0.64)
L-S	0.14	0.45 **	0.52 ***	0.84 ***	0.74 **	0.38	0.22	0.60
	(0.64)	(2.47)	(2.96)	(3.17)	(2.45)	(1.31)	(0.67)	(1.63)

el B: CAPM								
Long	0.48 **	0.47 ***	0.40 **	0.44 **	0.49 ***	-0.08	0.09	0.01
	(2.42)	(3.27)	(2.39)	(2.37)	(2.61)	(-0.42)	(0.45)	(0.05)
2	0.27 **	0.17 ***	0.12 *	-0.04	-0.06	-0.15	-0.18	-0.33
	(2.02)	(2.75)	(1.68)	(-0.76)	(-0.56)	(-1.30)	(-1.16)	(-1.63)
Short	0.21	-0.08	-0.24 *	-0.56 ***	-0.55 ***	-0.45 *	-0.31	-0.76 **
	(0.97)	(-0.47)	(-1.83)	(-4.06)	(-2.87)	(-1.80)	(-1.36)	(-2.24)
L-S	0.26	0.55 ***	0.63 ***	1.00 ***	1.04 ***	0.37	0.40	0.77 **
	(1.23)	(3.18)	(3.12)	(3.37)	(3.33)	(1.27)	(1.53)	(2.02)
nel C: FF3 al	lpha							
Long	0.35 **	0.37 ***	0.32 **	0.32 **	0.45 ***	-0.03	0.13	0.10
	(2.09)	(2.97)	(2.33)	(2.49)	(2.61)	(-0.17)	(0.63)	(0.44)
2	0.19 *	0.14 **	0.07	-0.05	-0.02	-0.12	-0.09	-0.20
	(1.85)	(2.50)	(1.31)	(-0.84)	(-0.15)	(-1.17)	(-0.64)	(-1.14)
Short	0.05	-0.17	-0.24 *	-0.52 ***	-0.38 ***	-0.29	-0.14	-0.43 *
	(0.30)	(-1.15)	(-1.79)	(-4.65)	(-2.59)	(-1.49)	(-0.62)	(-1.74)
L-S	0.30	0.54 ***	0.56 ***	0.84 ***	0.83 ***	0.26	0.27	0.53 *
	(1.42)	(3.03)	(3.35)	(4.50)	(3.65)	(0.98)	(0.92)	(1.76)

Table 4 Average returns and alphas of portfolios sorted by MISP and volume

This table reports the value-weighted average excess returns, CAPM alphas and FF3 alphas of portfolios double sorted on MISP and trading volume (TO). Long leg refers to the quintile with the lowest MISP and Short leg refers to the quintile with the highest MISP. L-S (H-L) refers to the long-short MISP (high-minus-low volume) portfolio spread. All portfolios are rebalanced monthly. Value-weighted monthly excess returns and alphas are reported in percentages and t-statistics are reported in parentheses. The sample period is from January 1980 to December 2021.

anel A: Ave	rage return							
	Low	2	Medium	4	High	M-L	H-M	H-L
	Volume	2	Volume	4	Volume	IVI-L	11-101	Π - L
Long	0.75 ***	0.85 ***	0.85 ***	0.84 ***	1.20 ***	0.11	0.35 *	0.45 **
	(4.01)	(5.33)	(4.77)	(3.98)	(4.45)	(0.71)	(1.74)	(2.18)
2	0.91 ***	0.85 ***	0.87 ***	0.63 ***	1.14 ***	-0.04	0.27	0.23
	(4.71)	(4.69)	(4.59)	(2.74)	(4.07)	(-0.27)	(1.64)	(1.04)
3	0.99 ***	0.78 ***	0.63 ***	0.92 ***	1.01 ***	-0.36 **	0.39	0.03
	(4.62)	(3.64)	(2.81)	(4.15)	(3.12)	(-2.27)	(1.47)	(0.08)
4	0.67 ***	0.76 ***	0.73 ***	0.60 **	0.70 **	0.06	-0.03	0.03
	(2.84)	(3.37)	(3.18)	(2.13)	(2.10)	(0.40)	(-0.13)	(0.13)
Short	0.22	0.46 *	0.32	0.03	-0.16	0.10	-0.48 *	-0.39
	(0.77)	(1.78)	(1.22)	(0.10)	(-0.40)	(0.53)	(-1.92)	(-1.24)
L-S	0.52 **	0.39 **	0.53 ***	0.81 ***	1.36 ***	0.01	0.83 ***	0.84 ***
	(2.56)	(2.00)	(3.07)	(3.97)	(5.56)	(0.04)	(3.56)	(3.33)

Panel B: CA	PM alpha							
Long	0.28 *	0.35 ***	0.26 ***	0.12	0.34 **	-0.03	0.08	0.05
	(1.93)	(3.14)	(2.72)	(1.21)	(2.57)	(-0.18)	(0.48)	(0.28)
2	0.41 ***	0.24 **	0.21 **	-0.14 **	0.18	-0.21	-0.02	-0.23
	(2.99)	(2.46)	(2.44)	(-1.98)	(1.42)	(-1.31)	(-0.14)	(-1.11)
3	0.51 ***	0.17	-0.07	0.12	-0.03	-0.58 ***	0.05	-0.54 **
	(2.83)	(1.19)	(-0.67)	(1.18)	(-0.18)	(-3.62)	(0.19)	(-2.02)
4	0.11	0.15	0.03	-0.26 *	-0.37 **	-0.07	-0.40 *	-0.47 *
	(0.55)	(1.14)	(0.21)	(-1.94)	(-2.28)	(-0.46)	(-1.68)	(-1.69)
Short	-0.41 **	-0.24	-0.46 ***	-0.92 ***	-1.38 ***	-0.05	-0.92 ***	-0.97 ***
	(-2.10)	(-1.28)	(-3.26)	(-6.43)	(-6.67)	(-0.27)	(-3.90)	(-2.98)
L-S	0.69 ***	0.59 ***	0.72 ***	1.04 ***	1.72 ***	0.03	1.00 ***	1.02 ***
	(3.57)	(2.99)	(4.52)	(5.53)	(7.67)	(0.12)	(4.48)	(3.82)

anel C: FF3	alpha							
Long	0.23 *	0.31 ***	0.23 ***	0.11	0.39 ***	0.00	0.16	0.16
	(1.93)	(3.03)	(2.64)	(1.27)	(3.11)	(-0.03)	(0.90)	(0.83)
2	0.30 **	0.21 **	0.16 *	-0.18 **	0.22 *	-0.14	0.06	-0.08
	(2.52)	(2.10)	(1.89)	(-2.46)	(1.77)	(-1.04)	(0.39)	(-0.41)
3	0.39 ***	0.07	-0.14	0.09	0.04	-0.53 ***	0.18	-0.35 *
	(3.06)	(0.62)	(-1.37)	(0.89)	(0.29)	(-3.91)	(0.89)	(-1.68)
4	-0.02	0.06	-0.06	-0.29 **	-0.28 *	-0.04	-0.22	-0.26
	(-0.13)	(0.55)	(-0.45)	(-2.24)	(-1.95)	(-0.29)	(-1.10)	(-1.19)
Short	-0.56 ***	-0.38 ***	-0.52 ***	-0.95 ***	-1.31 ***	0.05	-0.80 ***	-0.75 ***
	(-3.50)	(-2.61)	(-3.62)	(-6.73)	(-7.90)	(0.29)	(-3.97)	(-3.34)
L-S	0.80 ***	0.69 ***	0.74 ***	1.06 ***	1.70 ***	-0.05	0.96 ***	0.91 ***
	(4.68)	(3.96)	(4.86)	(5.49)	(8.28)	(-0.24)	(4.25)	(3.90)

Table 5 Investor attention and trading volume

This table reports the value-weighted investors' attention of portfolios sorted by trading volume (TO) in Panel A and value-weighted investors' attention of portfolios double-sorted by PEAD/FIN/MISP and trading volume (TO) in Panel B, C and D, respectively. Low/Medium/High volume refers to volume quintile 1/3/5. And M-L/H-M/H-L refers to medium-minus-low/high-minus-medium/high-minus-low portfolio difference in investors' attention. The attention measure is an average z-score of six individual attention proxies which include abnormal Google search volume, media coverage, abnormal EDGAR download, analyst coverage, price delay and abnormal Bloomberg download. All portfolios are rebalanced monthly, and t-statistics are reported in parentheses. The sample period is from January 1980 to December 2021.

Panel A: Investor attention	on sorted by TO							
	Low	2	Medium	1	High	M-L	н м	H-L
	Volume	2	Volume	4	Volume	IVI-L	H-M	П-L
Investor attention	0.15 *	1.08 ***	1.24 ***	1.23 ***	1.12 ***	1.10 ***	-0.12 ***	0.98 ***
	(1.92)	(14.68)	(28.86)	(29.84)	(30.77)	(14.36)	(-4.84)	(11.37)

1	-0.15 ***	0.73 ***	1.06 ***	1.12 ***	1.03 ***	1.21 ***	-0.03	1.18 ***
	(-3.76)	(10.67)	(25.54)	(26.67)	(28.08)	(25.96)	(-0.83)	(21.94)
2	0.08	1.09 ***	1.23 ***	1.24 ***	1.13 ***	1.16 ***	-0.10 ***	1.05 ***
	(1.21)	(16.77)	(30.76)	(31.64)	(28.49)	(15.77)	(-4.05)	(12.75)
3	0.12 *	1.15 ***	1.27 ***	1.25 ***	1.13 ***	1.15 ***	-0.14 ***	1.01 ***
	(1.66)	(17.83)	(32.25)	(31.11)	(30.42)	(15.68)	(-5.73)	(12.51)
4	0.11 *	1.07 ***	1.28 ***	1.26 ***	1.14 ***	1.17 ***	-0.14 ***	1.03 ***
	(1.88)	(14.35)	(29.08)	(31.54)	(32.93)	(16.29)	(-5.52)	(13.91)
5	-0.14 ***	0.69 ***	1.04 ***	1.09 ***	1.07 ***	1.18 ***	0.04	1.21 ***
	(-3.55)	(9.52)	(27.09)	(23.68)	(34.18)	(22.74)	(1.09)	(23.32)
nel C: Investor att	tention sorted by FIN	and TO						
1	0.05	1.18 ***	1.25 ***	1.23 ***	1.09 ***	1.20 ***	-0.16 ***	1.04 ***
	(0.66)	(14.35)	(19.46)	(18.21)	(22.00)	(13.46)	(-3.19)	(11.68)
2	0.26 ***	1.20 ***	1.25 ***	1.25 ***	1.14 ***	0.99 ***	-0.10 ***	0.89 ***
	(3.03)	(19.56)	(30.68)	(30.61)	(27.73)	(11.05)	(-5.42)	(9.45)
3	-0.12 ***	0.67 ***	1.03 ***	0.99 ***	1.05 ***	1.14 ***	0.03	1.17 ***
	(-2.95)	(12.65)	(17.51)	(21.39)	(40.30)	(19.77)	(0.55)	(27.31)

Panel D: Investor atte	ention sorted by MIS	SP and TO						
1	0.13	1.19 ***	1.33 ***	1.27 ***	1.14 ***	1.20 ***	-0.19 ***	1.01 ***
	(1.24)	(17.48)	(35.76)	(27.85)	(25.08)	(11.50)	(-6.73)	(8.49)
2	0.11 *	1.01 ***	1.23 ***	1.24 ***	1.15 ***	1.12 ***	-0.08 **	1.04 ***
	(1.78)	(13.26)	(28.89)	(29.48)	(29.27)	(15.42)	(-2.42)	(13.29)
3	-0.04	0.75 ***	1.13 ***	1.21 ***	1.14 ***	1.17 ***	0.01	1.18 ***
	(-1.40)	(10.87)	(23.74)	(29.95)	(30.84)	(24.78)	(0.33)	(26.61)
4	-0.13 **	0.59 ***	0.99 ***	1.07 ***	1.08 ***	1.13 ***	0.08 **	1.21 ***
	(-2.41)	(8.35)	(19.49)	(23.38)	(29.33)	(17.93)	(2.05)	(19.33)
5	-0.26 ***	0.39 ***	0.72 ***	0.93 ***	0.91 ***	0.98 ***	0.19 ***	1.17 ***
	(-5.87)	(5.61)	(11.24)	(17.78)	(26.03)	(13.07)	(3.86)	(21.20)

Table 6 Investor disagreement and trading volume

This table reports the value-weighted investors' disagreement of portfolios sorted by trading volume (TO) in Panel A and value-weighted investors' disagreement of portfolios double-sorted by PEAD/FIN/MISP and trading volume (TO) in Panel B, C and D, respectively. Low/Medium/High volume refers to volume quintile 1/3/5. And M-L/H-M/H-L refers to medium-minus-low/high-minus-medium/high-minus-low portfolio difference in investors' disagreement. The disagreement measure is the average z-score of three individual disagreement proxies which include stock return volatility, analysts' return forecast dispersion and analysts' earnings forecast dispersion. All portfolios are rebalanced monthly, and t-statistics are reported in parentheses. The sample period is from January 1980 to December 2021.

Panel A: Investor disagreer	ment sorted by 7	O						
	Low	2	Medium	Λ	High	M-L	H-M	H-L
	Volume	2	Volume Volume			IVI-L	П-1VI	11 - L
Investor disagreement	-0.98 ***	-1.03 ***	-0.93 ***	-0.61 ***	-0.04	0.06	0.89 ***	0.94 ***
	(-43.79)	(-33.15)	(-27.32)	(-13.87)	(-0.83)	(1.20)	(28.08)	(14.95)

1	-0.41 ***	-0.65 ***	-0.67 ***	-0.40 ***	0.16 ***	-0.27 ***	0.84 ***	0.57 ***
	(-9.73)	(-16.56)	(-23.10)	(-10.82)	(4.55)	(-5.95)	(25.89)	(12.69)
2	-0.91 ***	-1.03 ***	-0.93 ***	-0.63 ***	-0.07	-0.02	0.87 ***	0.85 ***
	(-40.35)	(-36.89)	(-28.36)	(-16.03)	(-1.50)	(-0.54)	(30.54)	(17.53)
3	-1.07 ***	-1.13 ***	-0.99 ***	-0.68 ***	-0.11 **	0.08 *	0.87 ***	0.96 ***
	(-42.03)	(-41.68)	(-29.80)	(-15.60)	(-2.11)	(1.80)	(26.81)	(14.16)
4	-0.96 ***	-1.08 ***	-0.95 ***	-0.66 ***	-0.14 ***	0.01	0.81 ***	0.82 ***
	(-35.29)	(-40.67)	(-29.06)	(-15.59)	(-2.81)	(0.20)	(25.38)	(13.83)
5	-0.41 ***	-0.65 ***	-0.70 ***	-0.44 ***	0.07 *	-0.28 ***	0.76 ***	0.48 ***
	(-8.78)	(-16.39)	(-31.31)	(-13.55)	(1.68)	(-5.65)	(20.85)	(7.18)
el C: Investor dis	agreement sorted by I	FIN and TO						
1	-1.01 ***	-1.08 ***	-0.93 ***	-0.69 ***	-0.24 ***	0.08 **	0.69 ***	0.77 ***
	(-28.40)	(-28.82)	(-24.65)	(-14.78)	(-5.33)	(2.04)	(19.48)	(20.16)
2	-1.04 ***	-1.08 ***	-0.90 ***	-0.59 ***	-0.07	0.14 ***	0.84 ***	0.98 ***
	(-48.55)	(-39.68)	(-26.39)	(-14.53)	(-1.51)	(3.04)	(29.98)	(16.88)
3	-0.71 ***	-0.78 ***	-0.71 ***	-0.36 ***	0.19 ***	0.00	0.89 ***	0.89 ***
	(-14.74)	(-21.00)	(-19.31)	(-9.63)	(5.38)	(0.03)	(25.14)	(19.09)

Panel D: Investor dis	agreement sorted by I	MISP and TO						
1	-1.04 ***	-1.13 ***	-1.00 ***	-0.75 ***	-0.28 ***	0.03	0.72 ***	0.75 ***
	(-35.80)	(-36.30)	(-27.01)	(-16.07)	(-6.07)	(0.55)	(21.98)	(11.68)
2	-0.97 ***	-1.01 ***	-0.92 ***	-0.63 ***	-0.14 ***	0.05	0.78 ***	0.83 ***
	(-28.75)	(-28.86)	(-25.49)	(-15.92)	(-2.84)	(1.21)	(24.77)	(14.16)
3	-0.91 ***	-0.87 ***	-0.85 ***	-0.57 ***	-0.03	0.05	0.82 ***	0.88 ***
	(-37.94)	(-22.90)	(-25.34)	(-14.07)	(-0.59)	(1.49)	(19.19)	(16.62)
4	-0.73 ***	-0.83 ***	-0.77 ***	-0.40 ***	0.11 **	-0.04	0.88 ***	0.84 ***
	(-18.65)	(-20.84)	(-25.31)	(-9.30)	(2.56)	(-0.89)	(19.66)	(15.87)
5	-0.35 ***	-0.47 ***	-0.49 ***	-0.21 ***	0.38 ***	-0.14 ***	0.87 ***	0.73 ***
	(-6.85)	(-8.18)	(-13.24)	(-5.07)	(8.96)	(-2.78)	(18.61)	(13.13)

Table 7 Fama-MacBeth tests on returns with trading volume, PEAD, attention and disagreement

The table reports the results of Fama-MacBeth regression. The dependent variable is the one-month ahead monthly excess return in percentage. The independent variables include PEAD quintile rank (PEAD), trading volume quintile rank (TO), Attention quintile rank (Attention), Disagreement quintile rank (Disagreement) and the interaction between PEAD quintile rank and trading volume quintile rank, the interaction between PEAD quintile rank and Disagreement quintile rank. We control firm size which is the logarithm of market capitalization and the logarithm of book-to-market ratio. Panel A reports the regressions results of full sample. We further divide the sample into low-volume state and high-volume state subsample based on trading volume quintile ranks that is the low-volume state subsample are stocks in trading volume quintile 3 to quintile 5. Panel B and Panel C report the results of low-volume state subsample and high-volume state subsample, respectively. Intercept and coefficients on controls are unreported for brevity. Newey-West robust t-values with four lags are reported in parentheses. *, **, and *** indicate significance at the 10%, 5%, and 1% levels respectively. The sample period is from January 1980 to December 2021.

Panel A: Full sample										
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
PEAD	0.225***				0.347***	0.355***	0.0404**	0.405***	0.167***	0.210***
	(13.32)				(13.26)	(13.75)	(1.98)	(13.65)	(6.39)	(6.58)
ТО		-0.0669			0.0580			0.00914	0.143***	0.108**
		(-0.95)			(0.74)			(0.12)	(2.87)	(2.18)
Attention			0.0795**			0.217***		0.189***		0.144***
			(2.39)			(5.16)		(5.61)		(4.36)
Disagreement				-0.106			-0.261***		-0.288***	-0.281***
				(-1.28)			(-3.00)		(-4.00)	(-3.89)
PEAD*TO					-0.0397***			-0.0287***	-0.0517***	-0.0448***
					(-4.70)			(-3.29)	(-6.28)	(-5.22)
PEAD*Attention						-0.0439***		-0.0312***		-0.0173**
						(-6.24)		(-4.42)		(-2.36)
PEAD*Disagreement							0.0529***		0.0640***	0.0603***
							(7.87)		(9.60)	(8.94)
Controls	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Average R-squared	0.0175	0.0286	0.0184	0.0321	0.0307	0.0205	0.0341	0.0325	0.0388	0.0404
Observations	1819736	1819736	1819736	1819736	1819736	1819736	1819736	1819736	1819736	1819736

	Pan	el B: Low-volu	me state subsar	nple	Pa	nel C: High-volu	ıme state subsaı	mple
	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)
PEAD	0.367***	0.419***	0.147***	0.184***	0.261***	0.343***	0.117**	0.184***
	(11.36)	(12.35)	(3.95)	(4.66)	(4.42)	(5.31)	(2.01)	(2.84)
ТО	0.252***	0.188**	0.323***	0.282***	-0.133	-0.174**	-0.0325	-0.0652
	(2.86)	(2.16)	(5.31)	(4.56)	(-1.51)	(-2.00)	(-0.50)	(-1.01)
Attention		0.181***		0.133***		0.204***		0.156***
		(6.01)		(4.71)		(4.44)		(3.47)
Disagreement			-0.306***	-0.300***			-0.283***	-0.266***
			(-4.44)	(-4.33)			(-3.33)	(-3.19)
PEAD*TO	-0.0478***	-0.0323**	-0.0576***	-0.0491***	-0.0209	-0.0109	-0.0362**	-0.0283*
	(-3.24)	(-2.05)	(-4.08)	(-3.24)	(-1.41)	(-0.73)	(-2.39)	(-1.85)
PEAD*Attention		-0.0324***		-0.0170**		-0.0369***		-0.0226**
		(-4.14)		(-2.16)		(-3.95)		(-2.37)
PEAD*Disagreement			0.0750***	0.0718***			0.0584***	0.0517***
			(8.92)	(8.53)			(6.43)	(5.67)
Controls	yes	yes	yes	yes	yes	yes	yes	yes
Average R-squared	0.0237	0.0255	0.0335	0.0351	0.0303	0.0332	0.0391	0.0417
Observations	1092154	1092154	1092154	1092154	1091534	1091534	1091534	1091534

Table 8 Fama-MacBeth tests on returns with trading volume, FIN, attention and disagreement

The table reports the results of Fama-MacBeth regression. The dependent variable is the one-month ahead monthly excess return in percentage. The independent variables include FIN tertile rank (FIN), trading volume quintile rank (TO), Attention quintile rank (Attention), Disagreement quintile rank (Disagreement) and the interaction between FIN tertile rank and trading volume quintile rank, the interaction between FIN tertile rank and Disagreement quintile rank. We control firm size which is the logarithm of market capitalization and the logarithm of book-to-market ratio. Panel A reports the regressions results of full sample. We further divide the sample into low-volume state and high-volume state subsample based on trading volume quintile ranks that is the low-volume state subsample are stocks in trading volume quintile 3 to quintile 5. Panel B and Panel C report the results of low-volume state subsample and high-volume state subsample, respectively. Intercept and coefficients on controls are unreported for brevity. Newey-West robust t-values with four lags are reported in parentheses. *, **, and *** indicate significance at the 10%, 5%, and 1% levels respectively. The sample period is from January 1980 to December 2021.

Panel A: Full sample										
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
FIN	-0.354***				0.0251	-0.427***	0.223***	-0.106	0.293***	0.182*
	(-4.75)				(0.31)	(-4.62)	(3.12)	(-1.18)	(3.06)	(1.71)
TO		-0.0730			0.173**			0.200***	0.0582	0.0855
		(-1.19)			(2.23)			(2.61)	(1.00)	(1.44)
Attention			0.0502*			-0.00485		-0.0796*		-0.0387
			(1.77)			(-0.10)		(-1.96)		(-0.95)
Disagreement				-0.0926			0.293***		0.263***	0.233***
				(-1.42)			(3.93)		(4.13)	(3.71)
FIN*TO					-0.106***			-0.124***	-0.0346	-0.0519**
					(-3.52)			(-4.11)	(-1.35)	(-2.00)
FIN*Attention						0.0261		0.0654***		0.0440**
						(1.25)		(3.28)		(2.22)
FIN*Disagreement							-0.173***		-0.156***	-0.143***
							(-5.56)		(-6.17)	(-5.62)
Controls	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Average R-squared	0.0201	0.0323	0.0196	0.0358	0.0356	0.0240	0.0390	0.0380	0.0445	0.0465
Observations	1495959	1495959	1495959	1495959	1495959	1495959	1495959	1495959	1495959	1495959

	P	anel B: Low-ve	olume state subsa	ample	Pan	el C: High-volu	ıme state subsan	nple
	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)
FIN	-0.110	-0.170*	0.133	0.0996	0.295*	0.116	0.463***	0.338*
	(-1.16)	(-1.68)	(1.16)	(0.88)	(1.79)	(0.64)	(2.76)	(1.79)
ТО	0.145	0.174	0.0791	0.0914	0.228**	0.226**	0.0444	0.0574
	(1.36)	(1.58)	(0.88)	(0.94)	(2.19)	(2.22)	(0.48)	(0.62)
Attention		-0.0334		-0.000356		-0.0613		-0.0141
		(-0.78)		(-0.01)		(-1.07)		(-0.24)
Disagreement			0.204***	0.198***			0.284***	0.250***
			(3.05)	(2.99)			(3.61)	(3.25)
FIN*TO	-0.0471	-0.0667	0.000593	-0.0103	-0.162***	-0.165***	-0.0652	-0.0739*
	(-1.05)	(-1.35)	(0.01)	(-0.23)	(-3.63)	(-3.75)	(-1.46)	(-1.68)
FIN*Attention		0.0402*		0.0219		0.0590**		0.0345
		(1.81)		(1.09)		(2.30)		(1.33)
FIN*Disagreement			-0.126***	-0.125***			-0.160***	-0.144***
			(-4.23)	(-4.26)			(-5.07)	(-4.52)
Controls	yes	yes	yes	yes	yes	yes	yes	yes
Average R-squared	0.0251	0.0275	0.0360	0.0381	0.0333	0.0370	0.0429	0.0461
Observations	897867	897867	897867	897867	897278	897278	897278	897278

Table 9 Fama-MacBeth tests on returns with trading volume, MISP, attention, and disagreement

The table reports the results of Fama-MacBeth regression. The dependent variable is the one-month ahead monthly excess return in percentage. The independent variables include MISP quintile rank (MISP), trading volume quintile rank (TO), Attention quintile rank (Attention), Disagreement quintile rank (Disagreement) and the interaction between MISP quintile rank and trading volume quintile rank, the interaction between MISP quintile rank and Disagreement quintile rank. We control firm size which is the logarithm of market capitalization and the logarithm of book-to-market ratio. Panel A reports the regressions results of full sample. We further divide the sample into low-volume state and high-volume state subsample based on trading volume quintile ranks that is the low-volume state subsample are stocks in trading volume quintile 3 to quintile 5. Panel B and Panel C report the results of low-volume state subsample and high-volume state subsample, respectively. Intercept and coefficients on controls are unreported for brevity. Newey-West robust t-values with four lags are reported in parentheses. *, **, and *** indicate significance at the 10%, 5%, and 1% levels respectively. The sample period is from January 1980 to December 2021.

Panel A: Full sample										
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
MISP	-0.326***				-0.224***	-0.462***	0.0899***	-0.343***	0.0799*	0.000184
	(-8.49)				(-5.89)	(-10.11)	(2.69)	(-7.99)	(1.76)	(0.00)
TO		-0.0702			0.0918			0.132**	-0.00688	0.0109
		(-0.99)			(1.35)			(1.99)	(-0.16)	(0.25)
Attention			0.0846**			-0.0351		-0.0826**		0.00299
			(2.53)			(-0.94)		(-2.30)		(0.10)
Disagreement				-0.113			0.385***		0.382***	0.356***
				(-1.36)			(5.21)		(6.29)	(5.96)
MISP*TO					-0.0340***			-0.0533***	0.00230	-0.00881
					(-2.76)			(-4.17)	(0.21)	(-0.80)
MISP*Attention						0.0445***		0.0596***		0.0307***
						(4.77)		(6.18)		(3.60)
MISP*Disagreement							-0.136***		-0.135***	-0.128***
							(-9.77)		(-10.41)	(-10.21)
Controls	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Average R-squared	0.0215	0.0282	0.0180	0.0317	0.0335	0.0245	0.0366	0.0355	0.0415	0.0432
Observations	1902176	1902176	1902176	1902176	1902176	1902176	1902176	1902176	1902176	1902176

	P	anel B: Low-vo	olume state subsa	imple	Par	nel C: High-volu	ıme state subsar	nple
	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)
MISP	-0.335***	-0.398***	-0.0438	-0.0670	0.0861	-0.0848	0.331***	0.226***
	(-7.85)	(-7.99)	(-0.94)	(-1.40)	(1.14)	(-1.09)	(4.34)	(2.87)
ТО	0.0869	0.110	0.0288	0.0233	0.187**	0.214**	0.0203	0.0431
	(1.09)	(1.47)	(0.52)	(0.43)	(2.21)	(2.53)	(0.34)	(0.71)
Attention		-0.0153		0.0521*		-0.123***		-0.0143
		(-0.41)		(1.74)		(-2.74)		(-0.35)
Disagreement			0.343***	0.337***			0.392***	0.350***
			(5.71)	(5.66)			(5.48)	(5.08)
MISP*TO	0.0180	0.00430	0.0421***	0.0386**	-0.103***	-0.119***	-0.0473**	-0.0591***
	(1.04)	(0.24)	(2.63)	(2.35)	(-5.19)	(-5.93)	(-2.56)	(-3.13)
MISP*Attention		0.0343***		0.0108		0.0697***		0.0339***
		(2.92)		(1.07)		(6.20)		(3.34)
MISP*Disagreement			-0.121***	-0.120***			-0.143***	-0.130***
			(-8.99)	(-9.41)			(-9.12)	(-8.46)
Controls	yes	yes	yes	yes	yes	yes	yes	yes
Average R-squared	0.0257	0.0278	0.0354	0.0371	0.0336	0.0367	0.0421	0.0447
Observations	1141607	1141607	1141607	1141607	1141004	1141004	1141004	1141004

Table 10 Fama-MacBeth tests on returns with trading volume, anomalies, and controls

The table reports the results of Fama-MacBeth regression. The dependent variable is the one-month ahead monthly excess return in percentage. Panel A reports the regression results on PEAD quintile rank (PEAD), trading volume quintile rank (TO) and their interactions. Panel B reports the regression results on FIN tertile rank (FIN), trading volume quintile rank (TO) and their interactions. Panel C reports the regression results on MISP quintile rank (MISP), trading volume quintile rank (TO) and their interactions. IVOL is standard deviation of residuals from Carhart four-factor regressions using the past month of daily return following Ang et al. (2006) from January 1980 to December 2021. Bid-ask spread is calculated following Corwin et al. (2011) from January 1980 to December 2021. VPIN is volume-synchronized probability of informed trading following Easley et al. (2012) from January 1993 to December 2013, thus we create VPIN_dummy which equal to 1 if VPIN is not missing and equal to 0 if VPIN is missing. IVOL, Bid-ask spread and VPIN are all converted into quintile rank. We control firm size which is the logarithm of market capitalization and the logarithm of book-to-market ratio in all the regressions. Intercept and coefficients on controls are unreported for brevity. Newey-West robust t-values with four lags are reported in parentheses. *, **, and *** indicate significance at the 10%, 5%, and 1% levels respectively.

Panel A: Full sample for PEAD					
-	(1)	(2)	(3)	(4)	(5)
PEAD	0.347***	0.144***	0.165***	0.355***	0.130***
	(13.26)	(5.14)	(5.24)	(13.20)	(4.00)
TO	0.0580	0.109*	0.0529	0.0550	0.0928
	(0.74)	(1.72)	(0.76)	(0.72)	(1.52)
PEAD*TO	-0.0397***	-0.0409***	-0.0326***	-0.0403***	-0.0411***
	(-4.70)	(-5.09)	(-3.93)	(-4.80)	(-5.09)
IVOL	, ,	-0.341***	, , ,	, ,	-0.279***
		(-6.16)			(-6.21)
PEAD*IVOL		0.0612***			0.0602***
		(8.66)			(6.90)
Bid-ask spread		` ,	-0.290***		-0.104**
-			(-5.01)		(-2.23)
PEAD*Bid-ask spread			0.0479***		0.00854
-			(6.95)		(1.01)
VPIN			` ,	-0.336***	-0.335***
				(-6.91)	(-6.98)
PEAD*VPIN				-0.00605	-0.00751**
				(-1.63)	(-2.09)
VPIN_dummy				1.290***	1.297***
_ •				(7.48)	(7.48)
Controls	Yes	Yes	Yes	Yes	Yes
Average R-squared	0.0307	0.0361	0.0363	0.0329	0.0412
Observations	1819736	1819633	1790037	1819736	1790011

Panel B: Full sample for FIN					
•	(6)	(7)	(8)	(9)	(10)
FIN	0.0251	0.361***	0.414***	0.0929	0.517***
	(0.31)	(3.56)	(3.72)	(1.12)	(4.41)
TO	0.173**	0.134*	0.192***	0.180**	0.161**
	(2.23)	(1.90)	(2.65)	(2.33)	(2.36)
FIN*TO	-0.106***	-0.0709***	-0.108***	-0.113***	-0.0917***
	(-3.52)	(-2.74)	(-4.01)	(-3.73)	(-3.52)
IVOL		0.188***			0.148***
		(4.13)			(3.15)
FIN*IVOL		-0.139***			-0.0997***
		(-5.76)			(-4.72)
Bid-ask spread			0.167***		0.0775
			(3.56)		(1.60)
FIN*Bid-ask spread			-0.117***		-0.0561**
			(-4.61)		(-2.35)
VPIN				-0.266***	-0.273***
				(-6.57)	(-6.40)
FIN*VPIN				-0.0151*	-0.0120
				(-1.78)	(-1.55)
VPIN_dummy				1.006***	1.009***
				(7.13)	(7.13)
Controls	Yes	Yes	Yes	Yes	Yes
Average R-squared	1495959	1495910	1475873	1495959	1475862
Observations	0.0356	0.0410	0.0410	0.0382	0.0465

Panel C: Full sample for MISP					
-	(11)	(12)	(13)	(14)	(15)
MISP	-0.224***	0.148***	0.200***	-0.246***	0.230***
	(-5.89)	(3.01)	(3.67)	(-6.23)	(4.11)
TO	0.0918	0.0839	0.141**	0.0751	0.0820
	(1.35)	(1.52)	(2.31)	(1.13)	(1.55)
MISP*TO	-0.0340***	-0.0234**	-0.0490***	-0.0304**	-0.0290**
	(-2.76)	(-2.08)	(-3.94)	(-2.50)	(-2.51)
IVOL		0.295***			0.272***
		(6.11)			(6.38)
MISP*IVOL		-0.128***			-0.109***
		(-10.43)			(-10.55)
Bid-ask spread			0.266***		0.0927***
			(5.68)		(2.79)
MISP*Bid-ask spread			-0.118***		-0.0440***
			(-9.50)		(-4.22)
VPIN				-0.398***	-0.410***
				(-7.57)	(-7.71)
MISP*VPIN				0.0169***	0.0176***
				(3.23)	(3.90)
VPIN_dummy				1.256***	1.278***
				(7.39)	(7.48)
Controls	Yes	Yes	Yes	Yes	Yes
Average R-squared	0.0335	0.0390	0.0394	0.0357	0.0441
Observations	1902176	1902051	1846827	1902176	1846796

Appendix

Table A1 Average returns and alphas of portfolios sorted by MOM and volume

This table reports the value-weighted average excess returns, CAPM alphas and FF3 alphas of portfolios double sorted on MOM and trading volume (TO). MOM is cumulative return over the past 12 months. Short leg refers to the quintile with the lowest MOM and Long leg refers to the quintile with the highest MOM. L-S (H-L) refers to the long-short MOM (high-minus-low volume) portfolio spread. All portfolios are rebalanced monthly. Value-weighted monthly returns and alphas are reported in percentages and t-statistics are reported in parentheses. The sample period is from January 1980 to December 2021.

	Low	2	Medium	4	High	MI	II M	11.1	
	Volume	2	Volume	4	Volume	M-L	H-M	H-L	
Short	0.20	0.32	0.33	-0.09	0.01	0.13	-0.32	-0.20	
	(0.66)	(0.94)	(1.07)	(-0.25)	(0.02)	(0.70)	(-1.11)	(-0.64)	
2	0.63 **	0.66 ***	0.80 ***	0.53 **	0.63 **	0.17	-0.17	0.00	
	(2.51)	(2.75)	(3.40)	(1.99)	(2.08)	(0.92)	(-1.20)	(0.01)	
3	0.65 ***	0.65 ***	0.74 ***	0.61 ***	0.70 **	0.09	-0.04	0.04	
	(3.17)	(3.60)	(3.89)	(2.71)	(2.34)	(0.65)	(-0.20)	(0.18)	
4	0.90 ***	0.93 ***	0.83 ***	0.79 ***	0.78 ***	-0.07	-0.05	-0.12	
	(4.64)	(5.13)	(4.14)	(3.55)	(2.61)	(-0.43)	(-0.26)	(-0.48)	
Long	0.94 ***	1.04 ***	0.95 ***	0.96 ***	1.18 ***	0.01	0.23	0.24	
	(4.16)	(4.42)	(3.81)	(3.94)	(3.48)	(0.07)	(1.12)	(0.88)	
L-S	0.74 ***	0.72 **	0.62 **	1.05 ***	1.17 ***	-0.11	0.55 **	0.44	
	(2.79)	(2.15)	(1.98)	(3.42)	(3.65)	(-0.43)	(1.97)	(1.40)	

Short	-0.53 **	-0.49 *	-0.58 **	-1.09 ***	-1.17 ***	-0.05	-0.59 **	-0.64 **
	(-2.37)	(-1.79)	(-2.54)	(-4.64)	(-5.13)	(-0.28)	(-2.07)	(-2.58)
2	0.07	0.03	0.11	-0.29 **	-0.36 ***	0.04	-0.47 ***	-0.44 **
	(0.36)	(0.19)	(0.78)	(-2.10)	(-2.72)	(0.21)	(-3.01)	(-2.00)
3	0.12	0.10	0.11	-0.16 *	-0.23 *	-0.01	-0.34 *	-0.35
	(0.70)	(0.84)	(1.03)	(-1.88)	(-1.73)	(-0.09)	(-1.78)	(-1.45)
4	0.42 ***	0.36 ***	0.19 **	0.04	-0.16	-0.23	-0.35 **	-0.58 **
	(2.63)	(3.55)	(1.99)	(0.47)	(-1.22)	(-1.56)	(-2.14)	(-2.89)
Long	0.38 **	0.39 **	0.23	0.19	0.21	-0.15	-0.02	-0.17
	(2.27)	(2.44)	(1.40)	(1.45)	(1.13)	(-0.80)	(-0.09)	(-0.66)
L-S	0.91 ***	0.88 ***	0.81 ***	1.28 ***	1.38 ***	-0.11	0.58 **	0.47
	(3.72)	(2.71)	(2.60)	(4.38)	(4.74)	(-0.43)	(2.01)	(1.54)

nel C: FF3 a	lpha							
Short	-0.64 ***	-0.61 ***	-0.71 ***	-1.21 ***	-1.24 ***	-0.07	-0.52 **	-0.59 **
	(-3.53)	(-2.59)	(-3.80)	(-5.61)	(-5.64)	(-0.36)	(-1.98)	(-2.31)
2	-0.07	-0.12	-0.02	-0.37 ***	-0.39 ***	0.05	-0.37 **	-0.32 *
	(-0.45)	(-0.84)	(-0.14)	(-2.81)	(-2.73)	(0.28)	(-2.18)	(-1.71)
3	0.00	0.04	0.05	-0.22 **	-0.21 *	0.05	-0.26 *	-0.21
	(-0.01)	(0.30)	(0.49)	(-2.51)	(-1.66)	(0.40)	(-1.66)	(-1.13)
4	0.33 **	0.33 ***	0.15	0.03	-0.14	-0.18	-0.29 *	-0.48 ***
	(2.39)	(3.11)	(1.60)	(0.34)	(-1.22)	(-1.38)	(-1.87)	(-2.76)
Long	0.33 **	0.39 ***	0.26 *	0.23 *	0.38 **	-0.07	0.12	0.04
	(2.22)	(2.66)	(1.76)	(1.86)	(2.34)	(-0.40)	(0.62)	(0.19)
L-S	0.98 ***	0.99 ***	0.98 ***	1.44 ***	1.61 ***	0.00	0.64 **	0.64 **
	(3.93)	(3.29)	(3.83)	(5.19)	(5.56)	(0.00)	(2.03)	(1.98)

Table A2 Alphas (equal-weighted) of portfolios sorted by PEAD/MOM and volume

This table reports the equal-weighted CAPM and FF3 alphas of portfolios double sorted on PEAD/MOM and trading volume (TO). Short leg refers to the quintile with the lowest PEAD/MOM and Long leg refers to the quintile with the highest PEAD/MOM in Panel A and Panel B, respectively. L-S (H-L) refers to the long-short PEAD/MOM (high-minus-low volume) portfolio spread. All portfolios are rebalanced monthly. Equal-weighted monthly returns are reported in percentages and t-statistics are reported in parentheses. FF3 alphas are reported in the last row of each panel. The sample period is from January 1980 to December 2021.

Panel A: CAPM and FF3 alphas of portfolios sorted by PEAD and TO Medium Low High 2 4 M-L H-M H-L Volume Volume Volume -0.31 * -0.45 ** -0.40 ** -0.62 *** -1.21 *** -0.09 -0.81 *** -0.89 *** Short (-1.70)(-4.93)(-3.24)(-2.59)(-2.39)(-3.99)(-5.10)(-0.60)0.23 0.10 0.11 -0.17 -0.56 *** -0.67 *** 2 -0.45 ** -0.12 (1.21)(0.68)(0.84)(-1.42)(-2.47)(-0.94)(-3.32)(-2.76)0.40 ** 0.08 -0.48 *** -0.26 ** -0.62 *** -0.88 *** 3 0.34 ** 0.14 (2.40)(1.06)(-2.36)(-3.43)(-3.69)(2.24)(0.66)(-2.72)0.58 *** 0.30 ** 4 0.46 *** 0.10 -0.39 ** -0.28 ** -0.68 *** -0.96 *** (3.25)(2.82)(2.25)(0.72)(-2.17)(-2.56)(-3.88)(-3.88)1.15 *** 0.97 *** 0.63 *** 0.23 -0.36 * -0.52 *** -0.99 *** Long -1.51 *** (6.00)(4.61)(3.37)(1.62)(-1.71)(-5.68)(-3.68)(-5.39)1.03 *** CAPM alphas 1.47 *** 1.42 *** 0.86 *** 0.84 *** -0.44 *** -0.19 -0.62 *** (13.58)(12.93)(9.86)(8.89)(7.04)(-3.16)(-1.43)(-4.14)-0.40 *** FF3 alphas 1.45 *** 1.44 *** 1.05 *** 0.90 *** 0.87 *** -0.18 -0.58 *** (13.05)(11.99)(10.73)(6.99)(7.45)(-3.09)(-1.48)(-4.04)

Short	-0.37	-0.45 *	-0.44 *	-1.02 ***	-1.51 ***	-0.08	-1.06 ***	-1.14 ***
	(-1.51)	(-1.90)	(-1.92)	(-4.34)	(-5.46)	(-0.45)	(-5.83)	(-3.84)
2	0.24	0.08	0.00	-0.21	-0.76 ***	-0.24 **	-0.76 ***	-1.00 ***
	(1.28)	(0.50)	(-0.02)	(-1.46)	(-4.38)	(-2.03)	(-4.40)	(-4.28)
3	0.48 ***	0.32 **	0.23 *	0.00	-0.52 ***	-0.25 **	-0.75 ***	-1.00 ***
	(2.86)	(1.97)	(1.68)	(0.01)	(-3.20)	(-2.07)	(-3.98)	(-4.14)
4	0.78 ***	0.58 ***	0.27 **	0.26 **	-0.23	-0.51 ***	-0.50 ***	-1.02 ***
	(4.66)	(3.66)	(2.09)	(2.13)	(-1.45)	(-5.51)	(-3.27)	(-5.04)
Long	1.00 ***	0.87 ***	0.66 ***	0.34 **	-0.14	-0.34 **	-0.80 ***	-1.14 ***
	(5.57)	(4.77)	(3.38)	(2.06)	(-0.65)	(-2.54)	(-4.31)	(-4.91)
CAPM alphas	1.37 ***	1.31 ***	1.10 ***	1.36 ***	1.37 ***	-0.27	0.27	0.00
	(7.20)	(6.82)	(5.02)	(5.61)	(4.94)	(-1.36)	(1.45)	(0.00)
FF3 alphas	1.41 ***	1.38 ***	1.19 ***	1.47 ***	1.52 ***	-0.22	0.32 *	0.10
	(7.29)	(7.32)	(5.49)	(6.06)	(5.71)	(-1.17)	(1.73)	(0.36)

Table A3 Alphas of portfolios with alternative sorting methods

This table reports the CAPM alphas of portfolios sorted by PEAD and trading volume with NYSE-breakpoints and with sequential sort in Panel A, the CAPM alphas of portfolios sorted by FIN and trading volume with NYSE-breakpoints and with sequential sort in Panel B, and the CAPM alphas of portfolios sorted by MISP and trading volume with NYSE-breakpoints and with sequential sort in Panel C. All portfolios are rebalanced monthly. Value-weighted monthly returns are reported in percentages and t-statistics are reported in parentheses. The sample period is from January 1980 to December 2021.

Panel A: CA	PM alpha of porti	folios sorted by P	EAD and TO					
	Low	2	Medium	4	High	M-L	H-M	H-L
	Volume	2	Volume	4	Volume	IVI-L	H-M	П-L
Sort with N	YSE breakpoints							
Short	0.09	-0.02	-0.02	-0.56 ***	-0.55 ***	-0.10	-0.53 ***	-0.64 **
	(0.59)	(-0.18)	(-0.14)	(-4.68)	(-3.26)	(-0.66)	(-2.67)	(-2.37)
2	0.19	0.13	-0.05	-0.07	-0.26 *	-0.24	-0.21	-0.45 **
	(1.56)	(1.36)	(-0.52)	(-0.63)	(-1.67)	(-1.58)	(-1.18)	(-2.06)
3	0.17	0.23 **	0.19 *	0.00	-0.21	0.02	-0.41 *	-0.38 *
	(1.57)	(2.37)	(1.71)	(0.00)	(-1.34)	(0.18)	(-1.71)	(-1.70)
4	0.25 **	0.07	0.13	-0.05	-0.17	-0.11	-0.30 *	-0.41 **
	(2.42)	(0.61)	(1.19)	(-0.51)	(-1.14)	(-0.82)	(-1.68)	(-2.11)
Long	0.40 ***	0.19	0.04	0.10	0.00	-0.36 *	-0.04	-0.40
	(2.88)	(1.25)	(0.36)	(0.93)	(0.02)	(-1.84)	(-0.19)	(-1.61)
L-S	0.31 *	0.22	0.06	0.65 ***	0.55 ***	-0.26	0.49 **	0.24
	(1.78)	(1.40)	(0.35)	(4.29)	(3.58)	(-1.12)	(2.43)	(1.02)

equential s	ort with PEAD fir	rst						
Short	0.26	0.59 **	0.61 **	0.44 *	0.42	0.35 *	-0.18	0.17
	(1.25)	(2.40)	(2.34)	(1.72)	(1.10)	(1.82)	(-0.78)	(0.52)
2	0.68 ***	0.76 ***	0.79 ***	0.66 ***	0.89 ***	0.11	0.10	0.21
	(3.04)	(3.78)	(4.35)	(2.85)	(2.82)	(0.68)	(0.52)	(0.88)
3	0.79 ***	0.86 ***	0.73 ***	0.88 ***	0.75 **	-0.06	0.02	-0.04
	(4.45)	(5.65)	(4.17)	(4.43)	(2.49)	(-0.37)	(0.09)	(-0.14)
4	0.88 ***	0.85 ***	0.78 ***	0.73 ***	0.91 ***	-0.09	0.13	0.04
	(4.12)	(4.55)	(3.92)	(3.63)	(3.05)	(-0.68)	(0.69)	(0.16)
Long	1.09 ***	1.18 ***	0.89 ***	1.02 ***	1.08 ***	-0.21	0.20	-0.01
	(4.58)	(5.48)	(4.06)	(3.99)	(3.15)	(-1.07)	(0.84)	(-0.04)
L-S	0.84 ***	0.58 ***	0.28 *	0.58 ***	0.66 ***	-0.56 **	0.38 *	-0.18
	(4.91)	(3.83)	(1.91)	(3.50)	(3.49)	(-2.41)	(1.86)	(-0.72)

	Low	2	Medium	4	High	M-L	Н-М	H-L
	Volume		Volume		Volume			
ort with N	YSE breakpoints							
Long	0.35 **	0.40 **	0.32 **	0.41 **	0.53 ***	-0.03	0.21	0.18
	(2.30)	(2.27)	(2.01)	(2.01)	(2.87)	(-0.18)	(1.07)	(0.87)
2	0.17 ***	0.15	0.10 *	-0.12 *	-0.06	-0.07	-0.16	-0.23
	(2.77)	(1.54)	(1.70)	(-1.68)	(-0.50)	(-0.86)	(-1.14)	(-1.52)
Short	-0.02	-0.11	-0.47 ***	-0.45 ***	-0.57 ***	-0.45	-0.10	-0.55 *
	(-0.10)	(-0.79)	(-3.08)	(-3.65)	(-2.99)	(-1.56)	(-0.47)	(-1.76)
L-S	0.37 **	0.51 ***	0.79 ***	0.86 ***	1.10 ***	0.42	0.31	0.73 **
	(2.06)	(2.89)	(2.77)	(2.92)	(3.63)	(1.12)	(1.06)	(2.00)
sequential s	ort with FIN first							
Long	0.48 ***	0.62 ***	0.24	0.33 *	0.47 **	-0.23	0.22	-0.01
	(2.82)	(3.66)	(1.48)	(1.84)	(2.37)	(-1.42)	(1.27)	(-0.06)
2	0.32 **	0.22 ***	0.11	0.00	-0.07	-0.21 *	-0.18	-0.40 **
	(2.38)	(2.87)	(1.43)	(0.04)	(-0.73)	(-1.82)	(-1.22)	(-2.08)
Short	0.03	-0.41 ***	-0.42 ***	-0.51 ***	-0.67 ***	-0.45 *	-0.25	-0.71 *
	(0.20)	(-3.54)	(-3.10)	(-3.18)	(-2.59)	(-1.91)	(-1.11)	(-1.90)
L-S	0.45 ***	1.03 ***	0.67 **	0.84 ***	1.14 ***	0.22	0.47 *	0.69 *
	(2.96)	(4.85)	(2.48)	(3.37)	(2.89)	(0.79)	(1.66)	(1.66)

Panel C: CA	PM alpha of portfo	olios sorted by MIS	SP and TO					
	Low	2	Medium	4	High	MI	II M	H-L
	Volume	2	Volume	4	Volume	M-L	H-M	Π - L
Sort with NY	SE breakpoints							
Long	0.32 ***	0.27 **	0.24 **	0.31 ***	0.34 **	-0.08	0.10	0.02
	(3.48)	(2.35)	(2.24)	(2.66)	(2.52)	(-0.70)	(0.54)	(0.12)
2	0.19	0.15	0.06	-0.02	0.28 **	-0.13	0.22	0.09
	(1.62)	(1.46)	(0.74)	(-0.16)	(2.04)	(-0.97)	(1.33)	(0.45)
3	0.24 **	0.06	-0.01	-0.21 *	-0.03	-0.25 *	-0.02	-0.27
	(2.37)	(0.51)	(-0.14)	(-1.83)	(-0.19)	(-1.69)	(-0.09)	(-1.38)
4	0.14	0.08	0.10	-0.11	-0.08	-0.04	-0.18	-0.22
	(1.06)	(0.57)	(0.84)	(-0.91)	(-0.45)	(-0.24)	(-0.83)	(-0.89)
Short	-0.14	-0.15	-0.36 **	-0.76 ***	-1.05 ***	-0.22	-0.69 ***	-0.91 ***
	(-0.88)	(-1.27)	(-2.55)	(-6.39)	(-5.70)	(-0.99)	(-3.88)	(-3.35)
L-S	0.45 ***	0.42 ***	0.60 ***	1.07 ***	1.39 ***	0.15	0.79 ***	0.94 ***
	(3.01)	(3.13)	(3.00)	(5.95)	(6.68)	(0.58)	(3.98)	(3.90)

equential s	ort with MISP firs	st						
Long	0.47 ***	0.35 ***	0.29 ***	0.03	0.35 ***	-0.17	0.06	-0.12
	(3.54)	(3.19)	(2.75)	(0.32)	(2.95)	(-1.18)	(0.35)	(-0.63)
2	0.47 ***	0.29 ***	0.22 **	0.00	0.04	-0.26 *	-0.18	-0.43 **
	(3.45)	(2.60)	(2.28)	(0.02)	(0.32)	(-1.79)	(-1.13)	(-2.18)
3	0.42 **	0.18	-0.06	0.08	-0.02	-0.48 ***	0.04	-0.44
	(2.30)	(1.18)	(-0.52)	(0.88)	(-0.13)	(-2.93)	(0.19)	(-1.64)
4	0.15	-0.02	0.03	-0.25 **	-0.37 **	-0.12	-0.40 *	-0.52 *
	(0.84)	(-0.13)	(0.24)	(-2.16)	(-2.21)	(-0.74)	(-1.79)	(-1.87)
Short	-0.45 **	-0.31 *	-0.65 ***	-0.90 ***	-1.56 ***	-0.20	-0.91 ***	-1.11 ***
	(-2.36)	(-1.80)	(-4.55)	(-5.54)	(-5.75)	(-0.96)	(-3.85)	(-2.90)
L-S	0.91 ***	0.66 ***	0.94 ***	0.93 ***	1.91 ***	0.03	0.97 ***	0.99 ***
	(4.67)	(3.52)	(4.56)	(4.55)	(6.17)	(0.10)	(4.40)	(2.88)

Table A4 Alphas of portfolios controlling size effect

This table reports the CAPM alphas of portfolios independently triple sort on PEAD (FIN or MISP), trading volume and size (using NYSE-breakpoints). All portfolios are rebalanced monthly. Value-weighted monthly alphas are reported in percentages and t-statistics are reported in parentheses. The sample period is from January 1980 to December 2021.

	Low	2	Medium	4	High	MI	II M	11.1
	Volume	2	Volume	4	Volume	M-L	H-M	H-L
Short	-0.45 ***	-0.24 *	-0.07	-0.43 ***	-0.79 ***	0.38 ***	-0.71 ***	-0.33
	(-2.75)	(-1.69)	(-0.62)	(-3.88)	(-4.61)	(2.73)	(-3.87)	(-1.30)
2	0.06	0.19 *	0.15	-0.10	-0.29 **	0.09	-0.44 ***	-0.36
	(0.33)	(1.67)	(1.62)	(-1.01)	(-2.10)	(0.63)	(-2.80)	(-1.35)
3	0.30 **	0.26 **	0.16	0.09	-0.30 **	-0.15	-0.46 ***	-0.61 ***
	(2.36)	(2.28)	(1.30)	(0.82)	(-2.39)	(-1.51)	(-2.62)	(-3.00)
4	0.36 **	0.34 ***	0.18 *	0.06	-0.19	-0.18	-0.37 **	-0.55 **
	(2.19)	(3.21)	(1.67)	(0.71)	(-1.37)	(-1.61)	(-2.25)	(-2.52)
Long	0.42 **	0.56 ***	0.33 ***	0.21 **	-0.09	-0.09	-0.43 **	-0.52 *
	(1.99)	(4.01)	(2.87)	(2.20)	(-0.54)	(-0.51)	(-2.06)	(-1.72)
L-S	0.88 ***	0.79 ***	0.41 ***	0.64 ***	0.69 ***	-0.47 **	0.29 **	-0.18
	(5.07)	(6.04)	(4.00)	(5.57)	(5.13)	(-2.41)	(2.01)	(-0.80)

Panel B: CAPM alpha of portfolios triple sort on FIN, TO and size Low Medium High M-L H-M H-L 2 4 Volume Volume Volume 0.53 *** 0.48 *** 0.41 ** 0.50 *** 0.34 * -0.12 -0.07 -0.19 Long (2.83)(3.31)(2.82)(1.77)(-0.80)(-0.40)(-0.87)(2.57)0.32 ** 0.26 *** 0.19 * -0.37 *** -0.50 *** 2 0.01 -0.18 -0.13 (-1.50)(2.33)(2.95)(1.88)(0.18)(-1.49)(-2.73)(-2.63)-0.69 *** -0.51 ** -0.94 *** Short 0.25 -0.05 -0.18 * -0.47 *** -0.43 * (1.11)(-0.31)(-1.72)(-3.61)(-3.80)(-1.88)(-2.51)(-2.80)0.53 *** 0.59 *** 0.97 *** 0.75 ** L-S 0.28 1.03 *** 0.31 0.45 ** (1.46)(1.19)(2.16)(4.04)(3.52)(3.72)(3.69)(2.02)

Panel C: CAPM alpha of portfolios triple sort on MISP, TO and size Medium Low High 2 4 M-L H-M H-L Volume Volume Volume 0.42 *** 0.46 *** 0.37 *** 0.31 *** 0.38 ** -0.05 0.01 -0.04 Long (3.17)(4.29)(2.91)(-0.21)(3.09)(2.51)(-0.53)(0.06)0.40 *** 0.34 *** 0.33 *** 0.10 0.11 -0.23 2 -0.07 -0.30 (2.91)(3.33)(2.91)(1.14)(0.78)(-0.63)(-1.64)(-1.55)0.55 *** 0.29 ** -0.41 *** -0.22 -0.63 ** 3 0.14 0.17 -0.07 (3.10)(-0.45)(-3.22)(-1.10)(-2.53)(1.99)(1.26)(1.53)-0.53 ** 4 0.15 0.17 0.01 -0.09 -0.37 ** -0.14 -0.39 ** (0.82)(1.29)(0.09)(-0.87)(-2.43)(-0.96)(-2.01)(-2.01)-0.32 -0.33 ** -0.51 *** -0.91 *** -1.43 *** -0.19 -0.93 *** -1.12 *** Short (-2.05)(-1.34)(-3.98)(-6.52)(-7.11)(-0.93)(-3.57)(-4.67)L-S 0.74 *** 0.79 *** 0.88 *** 1.82 *** 0.94 *** 1.08 *** 1.22 *** 0.14 (3.72)(5.03)(6.56)(6.52)(8.34)(0.65)(5.40)(4.07)